

The on-farm impacts of environmental flow rules in the Lachlan Valley





Report to the Lachlan River Management Committee

August 2001



The on-farm impacts of

environmental flow rules in the

Lachlan Valley

Report to the Lachlan River Management Committee

August 2001

Rohan Jayasuriya¹, Jason Crean¹ and Rendle Hannah²



¹ Economist, NSW Agriculture, Orange

² Agricultural Coordinator - Water Reform, NSW Agriculture, Orange

Office of Environment & Heritage LIBRARY - STACK

Shelf no: 333.913099449 JAY ID no: 20150049

Acknowledgments

The authors acknowledge the contribution of all those who have assisted with the completion of this study. Thanks particularly to members of the Socio-Economic Sub Committee of the Lachlan River Management Committee and members of various Local Consensus Data groups in the catchment for their contribution in eliciting the representative farm data. Thanks also to a number of NSW Agriculture staff for their contribution including Ian Smith, Irrigation Officer, Forbes and Rob Young (Program Leader, Economic Resource Policy). Thanks to Tahir Hameed and Shahadat Chowdhury (Hydrologists, DLWC) for the provision of simulated hydrology information.

Cover design by Alison Windus, Rural Womens Network, NSW Agriculture.

i

Table of contents

1. 1	Introduction	1
2.	The Lachlan Valley	2
2.1	Location	2
2.2	Irrigated agriculture	2
2.3	Reliability of irrigation supplies	5
3. 1	Environmental flows	8
3.1	Environmental issues	8
3.2	Environmental flow rules	8
4 . I	Methodology	10
4.1	An economic framework	10
4.2	Overview of approach	11
4.3	Model structure	12
2	4.3.1 Farm characteristics	12
4	4.3.2 DLWC Hydrology Model	17
2	4.3.4 Adjustment responses considered	18
5. <i>I</i>	Assessment of environmental flow rules	20
5.1	Evaluation of E73A and E131 environmental flow rules	20
5.2	Impact of E73A and E131 environmental flow rules in Dry Years	23
5.3	Assessment of previous options	25
6. 5	Summary	26
7. I	References	28
App	pendix 1: Representative farm data	29
App	pendix 2: Data used in the hydrology calculations	41
App	eendix 3: Hydrology Data from DLWC's IQQM Model	42
App	endix 4: Results of LRMC's E98 and E116 (previous) flow rules analysis	

List of Tables

Table 1: Enterprises in different zones of the Lachlan Valley	13
Table 2: Description of representative farms used in the analysis	14
Table 3: Results of LRMC's E73A and E131 flow rules analysis	21
Table 4: Results of LRMC's E73A and E131 flow rules analysis in Dry Years	24

List of Figures

Figure 1: Lachlan Catchment – Production Zones	3
Figure 2: Total area irrigated in the Lachlan Valley	4
Figure 3: Simulated Lachlan Valley January announced allocation percentages (1901-1997)	6
Figure 4: Actual announced allocations - initial and final allocations	7
Figure 5: Outline of Model Structure 1	2
Figure 6: LRMC's E73A and E131 flow rules – Impact on Farm Gross Margin	2

1. Introduction

River regulation and water extractions have contributed to a decline in the health of inland rivers across NSW. Environment flow policies introduced by the NSW Government in 1997 attempt to address this problem through reallocating a portion of water normally used by extractive uses back to the environment. The NSW Government developed interim environmental flow rules for a number of river systems to address river health needs, while keeping the impact on water users within 10 per cent of their average annual diversions. Water Management Committees (WMCs) on regulated river systems have reviewed the rules, and in most instances, have developed their own set of flow rules which continue to be adjusted as better information becomes available on their effects.

Environmental flows attempt to provide environmental benefits in the form of improvements in water quality and the health of natural ecosystems and aquatic biodiversity. These benefits may be achieved through the protection of low flows, providing triggers for fish and bird breeding events, mimicking natural flow variability and restoring a portion of freshes and high flows. The economic benefits attached to these environmental improvements may be significant. However, the economic trade-offs involved in obtaining environmental benefits may also be large.

The extent of trade-offs associated with establishing environmental allocations is an issue in the Lachlan Catchment. There are important river health issues in the catchment as well as a large irrigation industry dependent upon secure irrigation supplies. This study focuses on the likely impacts of environmental flows on broadacre irrigation farms³. The intention is to provide information and analyses to assist the Lachlan River Management Committee (LRMC) in its decision-making processes.

A combination of representative farm and hydrology simulation modelling is used to assess the impacts on agriculture from the implementation of different flow scenarios. These impacts are assessed by quantifying the difference in farm profitability between a base case (without environmental flows) and different environmental flow scenarios each involving reduced water availability. This study reports on the findings of that assessment and also complements an analysis being undertaken by DLWC on the wider regional socio-economic effects associated with changes in water management policy.

The general structure of the report is as follows. Section 2 contains an overview of irrigated agriculture in the Lachlan Valley. Section 3 describes the environmental flows proposed by the LRMC. Section 4 details the approach taken to assess on-farm agricultural impacts of the environmental flow rules. Section 5 presents the results of the analysis with respect to alternative management options, while Section 6 summarises the main findings of the report.

1

³ Environmental flows impact on general security rather than high security water allocations. Irrigation farms relying on general security water supplies are predominantly involved in broadacre agricultural production rather than permanent horticultural production.

2. The Lachlan Valley

2.1 Location

The Lachlan River Valley is located in Central Western NSW and covers an area of 84,700 square kms (DLWC, 1996). The Lachlan River begins on the slopes of the Great Dividing Range east of Gunning and flows north-west to Forbes and Lake Cargelligo through the Central Western Slopes and Plains. From Lake Cargelligo, the river flows south-west to the Great Cumbung Swamp where it occasionally joins the Murrumbidgee River (see Figure 1). Major tributaries of the Lachlan include the Abercrombie, Crookwell, Boorowa and Belubula rivers and Mandagery Creek which all join the Lachlan river above Forbes. Downstream of Forbes the main river channel diminishes and breaks down into a large number of effluent channels. The major water storages in the valley are Wyangala and Carcoar dams.

2.2 Irrigated agriculture

The Lachlan Valley is a significant agricultural area and much of the irrigation in the region is by licensed diverters from the Lachlan River. The only exception is the Jemalong Irrigation District which is a significant irrigation scheme lying to the west of Forbes. The Lachlan Valley has a licensed water entitlement of 665 GL (50 GL high security and 615 GL general security) although overall usage is usually around just 40-50 per cent of this. While the annual average allocation for general security licences has been in the vicinity of 80 percent over recent years (LIRAC, 1997), this is likely to decline as currently inactive licences are activated within the constraints of the Murray Darling Basin Commission (MDBC) Cap. The long-term cap diversion target for the Lachlan is 254 GL, just 40 per cent of licensed entitlement.

Irrigated agriculture in the Lachlan Valley is diverse. In the higher eastern area, upstream of the major storages, irrigated crops are dominated by horticulture and viticulture with some lucerne production in association with grazing enterprises. In the middle reaches of the Lachlan, the dominant users of irrigation water include canola, lucerne, maize, soybeans and horticulture as well as winter cereal crops. The lower reaches of the valley are dominated by summer crops such as maize and cotton; however, horticultural crops including citrus, viticulture, potatoes and others are also produced on suitable soils. This lower section of the valley is currently undergoing a rapid expansion in irrigated cropping (LIRAC, 1999).

The areas of crops and pastures irrigated from regulated supplies in the Lachlan catchment are shown in Figure 2. The areas relate to the 1999/2000 season and are taken from the crop return card data collected by DLWC at Forbes. The major crops are winter cereals, oil seeds and lucerne, each occupying around 17,000 - 20,000 hectares, and winter pasture and summer cereals with around 10,000 hectares each. There is about 5,000 hectares of summer pasture, around 2,000 hectares of vegetables and 1,500 hectares of winegrapes. The irrigation sector is an important contributor to the local economy with annual irrigated production valued at \$149 million (Donovan, 1998). Also, any economic impact on irrigated agriculture is likely to have flow-on impacts for regional income and employment. Similarly, many economic activities dependent on environmental quality, such as tourism and recreation, may also generate flow-on benefits for the regional economy.







Figure 2: Total area irrigated in the Lachlan Valley

The main features of irrigated agriculture in the Lachlan Valley can be best described with reference to five principal production zones moving from east to west (Figure 1). Soil types have been classified as either being light soils (loams, alluvial, self-mulching) or heavy soils (clays). The main irrigation technologies are surface irrigation (landformed and non-landformed), spray and trickle irrigation systems. The characteristics of these production zones are briefly outlined below.

Zone 1: Wyangala Dam to Payten's Bridge near Eugowra (including the Belubula River);

The main enterprises in this zone include vegetables (asparagus, tomatoes, sweet corn), winegrapes and lucerne (seed, hay and pasture). Spray irrigation is the dominant irrigation system. A mixture of soil types exists including deep alluvial, light red and loam soils. The licensed entitlement is around 82,000 megalitres with 11,000 hectares laid out to irrigation. It has the highest rainfall of any of the zones and the lowest temperature maximums. The mean annual rainfall for Cowra is 611 millimetres.

Source: Estimated from crop return card data collected by DLWC, 2001

Zone 2: Payten's Bridge to Island Creek off-take (above Condobolin);

In Zone 2, the main enterprises are wheat, canola, maize, lucerne (seed, hay and pasture), subclover and deciduous fruits (apples, peaches). Flood irrigation is dominant irrigation method. Deep alluvial soils dominate the zone. The licensed entitlement for Zone 2 is around 107,000 megalitres with 19,000 hectares laid out to irrigation. The mean annual rainfall for Forbes is 524 millimetres.

Zone 3: Island Creek off-take to Lake Cargelligo;

The main enterprises in Zone 3 are wheat, canola, maize, lucerne (seed, hay and pasture) and subclover. Flood irrigation is dominant in this zone and, most irrigation blocks are landformed. The main soil types in this zone are grey clays and deep alluvial soils. The licensed entitlement for region 3 is around 145,000 megalitres with 15,000 hectares laid out to irrigation. The mean annual rainfall for Lake Cargelligo is 425 millimetres.

Zone 4: Lake Cargelligo to Oxley;

In Zone 4, the main enterprises are wheat, canola, maize, lucerne (seed, hay and pasture), subclover and cotton. Flood irrigation is again dominant irrigation method. The main soil types in this zone are clays, loams and alluvials. The licensed entitlement for Zone 4 is around 206,000 megalitres, with 56,000 hectares laid out to irrigation. It has the lowest rainfall of any of the zones and the highest temperature maximums. The mean annual rainfall for Hillston is 361 millimetres. There has been a significant increase in irrigated agricultural output in Zone 4 in recent years, particularly in cotton production.

Zone 5: Jemalong Irrigation District

The main enterprises in Zone 5 are again wheat, canola, maize, lucerne (seed, hay and pasture) and sub-clover. Flood irrigation is dominant in this zone, with the majority of irrigation country landformed. The licensed entitlement for Zone 5 is around 80,000 megalitres with 42,000 hectares laid out to irrigation. The mean annual rainfall for the Jemalong Irrigation District is similar to that of Lake Cargelligo.

2.3 Reliability of irrigation supplies

Simulated hydrology data provided by DLWC (through IQQM Model) provides an indication of the reliability of irrigation supplies in the Lachlan Valley under historical climatic conditions, with current levels of development. Under base case conditions (without environmental flows), irrigators could expect to receive their full allocations or above in 62 per cent of years while they could receive less than 50 per cent allocation in 20 per cent of years (see Figure 3).

As with each regulated system within the State, the allocations provided to irrigators in the Lachlan depends upon the resources currently available in storage and those resources expected to be available during the season. An initial allocation made by DLWC at the commencement of the season is updated continuously to reflect rainfall conditions in the catchment. The allocation



Figure 3: Simulated Lachlan Valley January announced allocation percentages (1901-1997) (For the Base Case " C 71A" - without off-allocations)

assessment procedure is structured conservatively so that allocations will not need to be subsequently reduced during an irrigation season unless conditions realised are more severe than the worst recorded drought on record. As the period of record for critical streamflow statistics in most parts of NSW is around 100 years or so, the minimum-recorded streamflow sequence generally has about a 1 in 100 chance of occurring. That implies there is a 99 per cent chance that the announced allocations will not be reduced.

Not surprisingly, **actual** allocations announced at the start of the season generally have not been revised downwards (except in the worst recorded drought in 1982-83) since the introduction of volumetric allocations in the Lachlan in the early 1980's. Historical allocation announcements actually show that initial allocations were either set at their maximum level (100 per cent or higher) at the start of the irrigation season or set at a lower level and then considerably increased as the season progressed (Figure 4). Looking at those years where less than 120 per cent allocations were announced at the start of the irrigation season and excluding 1982-83 (drought year) and 1983-84 (initial announcement being zero), the average (15 out of 20 years) upward

revision in allocation was 41 per cent. The lower allocation levels⁴ experienced since 1995 reflect a mixture of both seasonal and policy influences, other than environmental flow rules (first introduced in 1998).





⁴ Drier seasonal conditions in the late 1990's resulted in lower storage levels in some years, and hence, less water available for allocation. The major policy influence was the introduction of MDBC Cap in 1995. Cap implementation in NSW involved the full recognition of 'sleeper and dozer' licenses in allocation announcements. Previously, under utilisation of licences acted to markedly inflate allocation announcements to regular water users.

3. Environmental flows

3.1 Environmental issues⁵

River regulation and water extractions have contributed to a decline in the health of the Lachlan River. Natural stream flows are usually highest from June to October and lowest in late summer. Regulation has changed this seasonal nature by capturing winter and early spring flows for release during summer. The average annual flow has been reduced at Oxley (at the end of the controlled system) from 234,000 ML/year under natural conditions to 120,000 ML/year currently. The catchment has experienced algal blooms, a decline in native fish species and an increase in exotic species, increasing river salinity and dryland salinity, a loss of native vegetation and declines in the health of wetlands.

River regulation has adversely affected native biota (particularly fish) and wetland ecosystems. Water released from the bottom of Wyangala and Carcoar dams is colder than natural flows, particularly in summer. There is evidence that these cold water releases have affected fish in the river. Water bird breeding in the Great Cumbung Swamp, Booligal Wetland and various other wetlands along the lower reaches of the river and fish migration and spawning events have been triggered by natural floods or freshes; but, regulation of flows may have caused more rapid recession of water levels and failure of breeding. Habitat maintenance of these areas is of environmental importance.

Salinity, in particular, represents a threat to agricultural productivity in the Lachlan if current trends are not reversed. Waterlogging and salinity in Jemalong Irrigation District, which is a significant irrigation scheme, has become a major problem, along with the disposal of water draining from irrigated land. These issues are currently being addressed through Land and Water Management Plans. Phosphorus concentrations generally increase as you move downstream and greatly increase during high flows as a result of stream bank and gully erosion due to land and riparian vegetation clearing. There have been substantial algal blooms in the lower section of the Lachlan River, including Lakes Brewster and Cargelligo.

3.2 Environmental flow rules

The Lachlan River Management Committee (LRMC) developed a set of flow rules for the 2000-01 season. These rules were designed to share water between users and the environment to improve river health by making provision for environmental water requirements, town water supplies, basic rights to stock and domestic water and identifying water availability for other extractive water uses, such as irrigation and other industries, thereby providing some level of water security to irrigators. The three individual flow rules adopted by the LRMC have been implemented as an integrated package, and consequently, should be viewed as simply attributes of the 2000-01 flow rules. A brief description of the flow rules is provided below⁶.

⁵ This section draws on material contained in EPA (1996) "Proposed Interim Environmental Objectives for NSW Waters".

⁶ The description provided draws on unpublished information by the DLWC Central West Region titled "Preliminary Draft Water Sharing Plan for the Lachlan Regulated River, 22/6/2001".

Rule 1: Translucent Releases

The translucent release rules have been designed to deliver flows to the effluents and wetlands in the lower system in a way which mimics natural flows. Translucent releases are to be made from Wyangala Dam during the June to November period to attain, in combination with tributary inflows, flows at Lake Brewster of 3,500 to 8,000 ML/day, depending on the storage level of the dam. The options looked at by the committee and used in this analysis involve variations in this translucent period. Water provided under this rule is classified as **Environmental Health Water**. In the Lachlan, this rule has been put in place to ensure that, to some degree, natural flow and variability is restored downstream of Wyangala Dam.

The volume of water that can be released under the translucency rule is limited to 350 GL. The actual amount released each year depends upon the storage level of Wyangala Dam.

Rule 2: Environmental Contingency Allowance

A 20 GL High Security Environmental Contingency Allowance (ECA) was established for management of critical environmental events such as: protection of bird breeding, native fish requirements and other threatened species; salinity dilution; and algal mitigation. Water provided under this rule can be classified as **Supplementary Environmental Water**.

In more recent options, the committee has added the concept of a Water Quality Allowance (WQA) to be used for problems such as blue green algae and high salinity levels, which might be addressed through dilution flows. The rules regarding the operation of the ECA and the WQA vary between the options considered in this analysis. Generally, the ECA will be eliminated during years when the 1st July allocation announcement plus percentage carryover for the Valley is below 50% and is not re-instated until allocation announcements plus percentage carryover for the Valley exceed 75%. The management of the ECA is specified as a number of sub rules relating to the treatment of unused ECA between seasons, and the use of ECA to supplement translucent releases.

Rule 3: Off Allocation

Off allocation water that might become available to irrigators is limited to 30,000 ML per annum. The remainder of these types of flows is reserved for the environment. Off allocation is only made available if Lakes Cargelligo and Brewster are guaranteed of filling and flows are in excess of the requirements for the environmental flow rules. Off allocation water is made available as a percentage of an individual irrigators entitlement and, when used, is not debited toward their annual allocation use.

4. Methodology

4.1 An economic framework

The implementation of environmental flow policies involves a re-allocation of resources. Such decisions are commonly assessed in a benefit-cost framework. The economic efficiency of different allocation policies can be assessed by comparing the social benefits and costs associated with each policy. There are however, a number of difficulties associated with adopting the standard benefit cost analysis framework when considering issues, which are likely to yield environmental benefits, like increased allocations to the environment. The major difficulty relates to the appropriate valuation of environmental benefits (particularly those in the non-use category) so that they can be incorporated into a benefit-cost framework.

To overcome some of the conceptual arguments regarding valuation, a variation on the standard benefit cost framework can be adopted through the use of an 'opportunity cost' or 'threshold value' approach. The threshold value approach avoids the need to directly place monetary values on environmental goods. The approach is based upon estimating the 'opportunity costs' which would be the consequence of a particular resource decision. In the case of environmental flows, the agricultural costs represent these opportunity costs. To gain a picture of the economic efficiency of environmental flows, these agricultural costs can be directly compared to the environmental outcomes (often quantified in non-monetary terms) which are expected from the proposal.

There are, however, further difficulties in applying a threshold value approach to community planning processes like the NSW water reforms. These relate principally to the broader interests of the community beyond economic efficiency. WMCs also consider whether water management changes are 'fair and reasonable', incorporating notions of equity between water users. Of key concern to many stakeholders is how the impacts of water management changes are distributed amongst different users. These users may be defined on a range of criteria including a geographic basis (eg. users in a specific part of a catchment) or a particular subset of users defined on water usage (eg. more active irrigators) or property or entitlement sizes (eg. small users).

The evaluation of agricultural impacts therefore requires analysis at two levels. First, at a broader regional scale, agricultural impacts can be assessed, and subsequently used, in a threshold value approach to determine the overall economic efficiency of options. Second, impacts on a more disaggregated basis can be assessed to provide WMCs with distributional information on how subsets of the population might be affected. The first issue is addressed through wider regional effects of changes in irrigated production⁷, being undertaken by DLWC that also complements this study. NSW Agriculture has undertaken economic assessments at the farm level. The on-farm impacts provide distributional information and are the focus of this study.

⁷ NSW Agriculture has also previously undertaken some regional analysis of agricultural impacts. See NSW Agriculture, 1996 and 1998.

4.2 Overview of approach

There is a broad range of techniques available for assessing the farm level impacts of water policy reforms. These techniques range from simple budgeting methods to formal optimisation models. The applicability and appropriateness of any of these techniques depends ultimately on the context of the analysis, the problem being addressed and the nature of the farming systems under consideration.

This study used a representative farm modelling approach to evaluate the on-farm impacts of environmental flows. A standard whole farm budgeting framework was adopted to consider changes in water availability and associated farm adjustment responses. This framework is used to assess the impacts that reduced water availability has on the profitability of representative irrigation farms involved in broadacre agricultural production. The impacts of environmental flows were estimated across a range of different years utilising historical weather data and simulated hydrology data.

A critical component of the approach is the selection of representative farm types. The socioeconomic sub-committee of the LRMC identified six representative farm types to depict the main farming systems in 5 principal production zones (see Figure 1) moving from east to west in the Lachlan Valley. It is recognised that these representative farms do not represent the full diversity of surface water use in the region, however, they do represent the predominant broadacre irrigation farming systems relying on surface water in the catchment.

Representative farm models were developed to capture the nature of these six irrigation-farming systems identified. The models are set out as whole farm budgets and explicitly model irrigation requirements and water availability over a 97-year simulation period. Irrigation requirements are driven by fluctuations in rainfall availability with monthly crop evapo-transpiration requirements fixed. DLWC hydrology simulation information (from IQQM Model) is used to represent irrigation water availability over the same period. The economic models are solved on the basis of annual allocation availability under different environmental flow scenarios, expressed as a percentage of licensed entitlement. Figure 5 shows a schematic of the structure of the model.

Impacts of environmental flows on farm profitability are assessed in terms of whole farm gross margin and net farm income. Definitions of these profitability indicators are as follows:

- Whole farm gross margin sum of individual enterprise gross margins (enterprise income less enterprise variable costs) received from all farm enterprises;
- Net farm income whole farm gross margin less overhead costs (overhead costs include depreciation, administration, permanent labour and rates but exclude finance costs like interest and rent on leases);

Effects on whole farm gross margin and net farm income essentially measure the impacts on the income generation capacity of the representative farms. Whole farm gross margin aggregates the contribution of each farm activity and gives an indication of returns prior to the consideration of overheads or fixed costs of the farm. Net farm income is a measure of farm profit and measures the return to the operator for their labour and management and the return to all capital invested in the farm whether it is borrowed or not. Because net farm income excludes finance costs, comparisons of results are not complicated by differences in the level of indebtedness peculiar to

Methodology

particular farms. More details of the major components of the modelling approach (as outlined in Figure 5) are provided below. Full details of data used in the solution process including farm details, evapo-transpiration requirements, effectiveness of rainfall and irrigation efficiencies etc, are presented in Appendices 1 - 2.





4.3 Model structure

4.3.1 Farm characteristics

NSW Agriculture developed whole farm models to represent the key characteristics of irrigation farming in 5 principal production zones moving from east to west in the catchment. Data on the key features of representative farms were collected using a local consensus data (LCD) approach. The LCD technique is a way of quickly obtaining data on the structure of farming for a particular farming system in a particular locality. In general terms, the approach involves a meeting between a small group of experienced farmers and officers of the Department of Agriculture to discuss all the practices which have a bearing on the costs and returns of a 'typical' farm in the area of interest. As discussion proceeds, a consensus of opinion or agreement is reached on the size and

nature of the 'typical farm' and on relevant aspects of farm production. These relate to crop areas and yields, prices, management practices, water use, variable costs and overhead costs, and as a consequence, net farm income. The information provided is cross-checked against existing sources for consistency.

It is important to note that the approach is not statistically based, and as a consequence, is not truly representative of farms on the basis of any single characteristic. However, the approach does draw on available statistical data (farm size, irrigation allocations, crop areas etc) as a way of putting bounds around what might be considered as typical farms by participants. Ultimately, the figures reported through this technique are not average figures, but typical figures for farmers represented by the group. This has both advantages and disadvantages. An advantage is that typical figures for a targeted group can be more representative than average figures, which are commonly distorted by various sources of sampling error arising from variability in the survey population. A significant disadvantage is that figures derived cannot be simply aggregated up for use in a regional analysis. A more detailed discussion on the technique and its relative advantages and disadvantages can be found in Jayasuriya, Catt & Young (1999).

The LCD meetings were conducted in July-August 1999 in various parts of the catchment. The LCD groups consisted of practicing farmers of the area, as nominated by the socio-economic sub committee of the LRMC and NSW Agriculture staff. The range of irrigated farming systems identified by the socio-economic sub committee and later endorsed by the various LCD groups in relevant zones, are shown in Table 1. The highlighted enterprises shown are those farm types selected for representative farm modelling. The main physical characteristics of these representative farms in terms of property sizes, water entitlement and usage, breakdown of irrigated and dryland enterprise and financial characteristics are given in Table 2.

Zone	Number of Licences	Enterprises	Number of Farms
Zone 1	307	lucerne, and fat lamb production sweet corn combined with lucerne & mixed farming horticulture (fresh vegetables / vines)	120 20 5
Zone 2	412	lucerne, grazing & winter crops (mixed farms) orchards dairy	150 15 10
Zone 3	320	grazing / winter crops (small & large farms 50% each) graziers / water traders summer crops	100 50 10
Zone 4	340	graziers / water traders cotton & summer crops horticulture (citrus / vines / fresh vegetables)	100 30 10
Zone 5	1	lucerne & mixed farming summer crops (maize) grazing / winter crops	45 30 30

Table 1: Enterprise.	in different zones	of the Lachlan Valley
----------------------	--------------------	-----------------------

Methodology

	Zone 1	Zone 2	Zone 3 Small Farm	Zone 3 Large Farm	Zone 4	Zone 5
Key physical characteristics						
Total farm size (Ha)	304	800	1000	5000	7500	2000
Area set up for irrigation (Ha)	152	320	200	200	1300	750
Water entitlement (ML)	600	1000	972	972	4000 ⁸	1400
Estimated Av. Water use ⁹	471	531	743	525	4937	1373
Irrigated enterprises (Ha)						
Irrigated Wheat	15	40	75	20	60	50
Irrigated Oats			25	20		
Irrigated Canola	19	20				50
Irrigated Cotton					250	
Irrigated Maize					180	100
Irrigated Lucerne Hay	76	80	50			
Irrigated Perennial Pasture	1				60	100
Irrigated Annual Pasture		20	50	160		50
Dryland enterprises (Ha)						
Wheat	61	160	200	1000	300	300
Barley						100
Oats				200		100
Canola	19	160	100	300		250
Lucerne Hay (establishment)	19	20				
Improved / Perennial Pasture	95	280	400	2000	5900	800
Fallow / developing / non-arable		20	100	1300	750	200
Number of Sheep	700	1500	1200	3200	3000	1700
Number of Cattle		50	40	200	50	150
Farm labour (Weeks)						
Owner / Manager & Family	50	50	50	100	100	50
Permanent Labour		48	48	48	48	48
Casual Labour	25	25		48		10
Average farm performance ¹⁰						
Whole Farm Gross Margin	92,828	162,062	111,868	231,127	443,529	159,943
Total Operating Overheads	54,446	99,153	60,253	140,629	204,946	120,197
Net Farm Income	38,383	62,909	51,615	90,498	238,584	39,747
Business Return	-4,167	8,209	13,515	7,998	165,234	-3,953
Total Assets	1,315,700	2,285,500	1,541,500	3,490,000	4,038,606	3,507,500
Total Liabilities	230,000	400,000	220,000	500,000	750,000	450,000

Table 2: Description of representative farms used in the analysis

⁸ Average water use exceeds the surface water entitlement because this representative farm also holds a 2,000 ML groundwater base entitlement and a 2,000 ML conjunctive use entitlement. Surface water supplies are utilised initially by the farm with groundwater used only to supplement surface water availability.

⁹ Calculated through model runs for the 97-year (full simulation period) monthly average rainfall.

¹⁰ Simulated farm performance based on 97-year monthly average rainfall.

A brief description of each of the farms is provided below.

- i) **Zone 1 –Lucerne and fat lamb production.** This is seen as the 'traditional' type mixed farm of the Cowra to Forbes region, representing approximately 120 irrigators. The farm type has an average area of 300 Ha (range 150 to 600 Ha), with 600 ML (range 150 to 1,000 ML) water entitlement. Average water use is about 80%. The farm is run by the owner/manager, with employment of casual labour during peak demand periods adding up to about a half time employee. Lucerne is the main focus of irrigation, and used for hay and/or fattening stock, depending on market conditions and the availability of water. Sale of hay (or hay products) is the major contributor to income. Winter crops, largely wheat and canola are grown, with some irrigation when necessary and available. A cereal crop for feed (grazing and grain) is also generally part of the farm program. From a gazing perspective, this type of farm was the only one not to include cattle, with the focus on fat lamb production.
- ii) Zone 2 Lucerne, grazing and winter Crops (mixed farming). This farm type is also included as a 'traditional' Lachlan farm, most common between Eugowra and Condobolin. It is estimated to represent about 150 irrigators, with an average area of 800 Ha (range 600 to 1,400 Ha) and 1,000 ML (range 600 to 1,800 ML) water entitlement. Average water use is about 50% of entitlement. This type of farm is seen as a typical family farm, often run by a father and son team, with the use of casual labour during peak demand periods adding up to about a further half time employee. Lucerne is again the main focus of irrigation, and used for hay and/or fattening stock, depending on market conditions and the availability of water. Winter (wheat & canola) cropping, both irrigated and dryland, plays a larger role than the zone 1 type farm. Irrigated annual pasture also plays an important role for stock feed, particularly in the dry years this is especially important for some with Stud enterprises.
- iii) Zone 3 Grazing/winter crops ("small" farm type). This type is classified as "small", as it generally includes ownership of a single 'river' property which includes an area laid out to irrigation. Again, it is seen as a 'traditional' Lachlan farm, with a focus on fat lamb production, and winter crops. Farms of this type are typically found between Forbes and Lake Cargelligo, including the Jemalong Irrigation District, and represent about 65 irrigators. Farm size averages 1,000 Ha (range 400 to 5,000 Ha) with 972 ML (range 400 to 8,000 ML) water entitlement. Average water use is about 80% of entitlement. This type of farm is also seen as a typical family farm, often run by a father and son team (two labour units). Irrigation is primarily used for fodder and pastures crops (including lucerne), with returns from sale of hay being an important component. Wheat, and probably a growing use of Canola, are important parts of the farm rotation, both irrigated and dryland.
- iv) Zone 3 Grazing/winter crops ("large" farm). This type is classified as "large", as it includes ownership of a significant 'off-river' dryland block, as well as the 'river' block with irrigation. The main focus of this farm is dryland winter crops, although in the past it may have been primarily a grazing property. Again, farms of this type are typically found between Forbes and Lake Cargelligo, and represent about 65 irrigators. Farm size averages 5,000 Ha, with a 1,000 Ha 'river block', similar to the "small" farm in (iii) above and dryland being 4,000 Ha (overall range 2,000 to 20,000 Ha). Generally water

entitlements are again 972 ML, with a range of 400 to 8,000 ML. Irrigation is an important adjunct to this farm type, being particularly important in dry periods. With this emphasis on reliability rather than use, average water use is about 50% of entitlement. The farm requires two family labour units, as well as two full time employees. Irrigation is primarily used for pastures, which might be an important part of a stud enterprise.

- Zone 4 Cotton and summer crops. This type of farm represents the industry that v) has been providing the largest growth in the Lachlan Valley over the last decade. Due to the climatic requirements of current cotton varieties, it is located west of Lake Cargelligo, mostly based around Hillston. It is estimated to represent about 30 farms to date, and is the only type in this study to include groundwater as a significant part of irrigation. Farm size averages 7,500 Ha, with a range of 2,000 to 50,000 Ha, reflecting the historically large grazing properties on which irrigation has been developed. Surface water entitlements average 4,000 ML, with 2,000 ML base plus 2,000 ML conjunctive groundwater licences. To represent the variation in irrigation systems of the area, this farm has 100 Ha under spray irrigation, with 1,000 Ha landformed for flood irrigation, and a further 200 Ha being developed. Whilst cotton is the main irrigation crop, maize, cereals and lucerne based pasture are important part of irrigation rotations. There is potential for other summer crops. A merino based 3,000 sheep flock and a 50 cow cattle enterprise make up the stocking component, mostly run on the dryland area. Labour units include two family members and a full time employee.
- vi) Zone 5 Summer crops (maize). This type of farm is seen as the main intensive irrigation type of the Jemalong Irrigation District (Zone 5), although it also represents farms outside the district, but generally between Forbes and Lake Cargelligo. It is estimated to represent about 30 farms. Maize has been the most common summer crop, but the search is on for profitable enterprises, which might complement this. Because of the fixed costs of Irrigation District, the cost per ML of water increases with lower use. This provides an additional incentive for the search for profitable summer crops, which remains elusive. Wheat, canola and pastures (lucerne, medic and sub clover) are part of the irrigation rotation. Dryland cereals and pasture also make a major contribution. From the grazing perspective, there is a strong emphasis on cattle combined with a traditional merino flock. Farm size averages 2,000 Ha (range 600 to 5,000 Ha) with 1,400 ML water entitlement (range 800 to 2,600 ML). Average water use is about 95% of entitlement. Labour units include the owner/manager, a full time employee, and some casual labour at the busy times.

4.3.2 DLWC Hydrology Model

The DLWC IQQM model simulates the operation of the Lachlan system by calculating the monthly announced allocation percentages and total allocation diversions for each year of a 97 year simulation period from 1901 to 1997. The model is set to represent, as closely as possible, all the factors affecting water use as they were in 1993-94. These factors include dams and water storages then in place, the water allocation rules, amount of land being irrigated, the year by year planting decisions made by farmers etc. The model is simulated with the actual rainfall, evaporation and water inflow for the period 1901 to 1997 to obtain the simulated hydrology output. This hydrology simulation approach has been used in other issues including the analysis of river flow objectives (EPA, 1996; NSW Agriculture, 1996 & 1998; DLWC, 1998) and the Snowy Water Inquiry, 1998.

Hydrology simulation information from DLWC's IQQM model was used to represent the allocations that irrigators were expected to receive under different environmental flow scenarios through time (see Appendix 3). The economic modelling uses this hydrology data as input into the extent of irrigated crops grown in each of the representative farms. However, the historically conservative nature of allocation announcements by DLWC (as discussed in Section 2) suggests that farmers would be unlikely to base their farm plans solely on announced allocations at the beginning of the season (August). The extent that irrigators would upwardly revise allocation announcements depends on irrigators' attitude to risk, which is likely to be individual specific. For the purposes of the analyses, it was assumed that irrigators would plan to receive an allocation, which approximated that of the January announcement in each year of the simulation period. As a consequence, the study is assuming that irrigators are well informed about the usual increase in allocation announcements and that they base their crop planting decisions on higher water availability than that is actually announced at the start of the irrigation season¹¹.

4.3.3 Historical weather data

The irrigation requirements of different crops for each representative farms is simulated over a 97 year period based on historical weather data in the Lachlan Catchment. Depending on the location of the representative farm, the monthly rainfall is obtained from four different rainfall stations in the catchment namely Cowra, Forbes, Lake Cargelligo and Hillston. The effectiveness of rainfall data provides information on the contribution that rainfall makes to meet crop evapo-transpiration requirements. The data is provided on a monthly basis. The crop evapo-transpiration data was sourced from Jemalong Land and Water Management Planning evaluations and other NSW Agriculture reports (NSW Agriculture, 1996 and 1998). These are provided by crop and by month and are fixed during the simulation.

¹¹ Most farmers would be well aware of the tendency for actual announced allocations to increase through the year. For example, according to **historical** announced allocations (see Figure 4), the average increase in allocations over the season between 1982 and 2001 was 41 per cent (excluding two dry years and years when the maximum of 120% allocation was announced at the start of the irrigation season). The **simulated** hydrology for the base case C71A (see Appendix 3) shows a 29 per cent increase from August to January on average across the full simulation period.

4.3.4 Adjustment responses considered

The types of adjustment responses taken by farmers in response to lower water availability will have a significant bearing on the magnitude of effects. Some responses can be undertaken in the short term (temporary trade in water, change enterprise mix) while other responses require a longer-term time frame (investment in irrigation infrastructure to improve irrigation efficiency). The focus of this study is on the shorter-term responses that farmers can make to lower water availability. The responses considered are described below.

Adjustment response 1: Buy/sell water in the temporary market

The purchase of surface water was supported by the LCD groups who indicated that a number of irrigators trade in the temporary water market. The purchase of water can be assessed within the modelling framework through the simulation of annual water demands based on climatic conditions. When water demand exceeds availability in a particular year, the model purchases the deficit on the temporary transfer market and maintains the original enterprise levels. The price at which the model purchases surface water is determined by an aggregate surface water demand function for the Lachlan Valley¹². The price of water varies annually depending on the announced allocations for the surface water users. The price ranged from \$15 to \$80 with an average of \$23 per ML across the 97-year period. These prices are consistent with the views of members of the LCD group who discussed the possibility of buying surface water for around \$20/ML. The model also assumes that any surplus water could be sold at the same price as determined above. For the Zone 4 representative farm, in addition to the above, supplementation with groundwater is possible too.

Adjustment response 2: Change enterprise mix

The purchase of additional surface water to offset reductions in water supplies may not be the most feasible adjustment response for some irrigators. Only some water users access temporary traded water, and there are also restrictions in place for the transfer of water between license holders. An alternative response for some irrigators may involve making changes to enterprise mix and crop rotations to accommodate water shortages resulting from environmental flow policies. This adjustment response is incorporated into the analysis by progressively reducing low priority irrigated enterprises, as identified by the LCD group, and replacing sacrificed irrigated enterprises with dryland alternatives to partially offset the loss of income. The priority of the farm enterprises is determined on the basis of the gross margin per ML of water use. For all representative farms, dryland barley is included as a replacement crop.

Consideration of both responses together

Sometimes it is not rational to buy water under adjustment response 1, when the marginal value of water (the value of agricultural production with an additional ML of water) is lower than the

¹² The function is derived through the application of NSW Agriculture's regional linear programming models, which attempt to represent surface irrigated agriculture across the same five irrigation zones in the Lachlan Valley. An existing economic model of the Jemalong area was initially compiled by Randall Jones and Anthea M'cClintock, formerly of the Economic Services Unit of NSW Agriculture. The larger LP model of the Lachlan was extended by Randall Jones, Jason Crean and Margot Fagan and has been further revised by Rohan Jayasuriya and Jason Crean. Ian Smith, Irrigation Officer, Forbes has provided substantial technical input.

market price of water. Under such circumstances it becomes more profitable to cut back low value enterprises as under the adjustment response 2. Similarly, sometimes it is not rational to cut back enterprises under adjustment response 2, when the marginal value of water is higher than the market price of water and when additional water is available to the farm on the temporary transfer market. This problem is unavoidable in an approach where the two adjustment responses are considered as two mutually exclusive events in a simulation period.

To avoid these problems, this study adopts an approach to choose between the two adjustment options based on their relative financial merits. The model makes a rational decision (whether to buy water or cut back enterprises) based on an overall goal of maximising farm returns. Under this approach, the marginal value of water is the driving force that determines which adjustment response should be adopted. Consequently, each year the model determines whether the farm should purchase temporary trade water, use groundwater if available, or change enterprise mix to make the best use of the available water. The model is based on linear programming techniques and attempts to maximise farm gross margin (M) according to the objective:

$$M = \sum_{j=1}^{n} (c_j - a_{ij} \cdot x_j \cdot p_i), \qquad (j = 1, \dots, n)$$

Where:

 c_i denotes all the revenue from activities j;

 x_j is the magnitude of activity j;

a_{ij} is the amount of resource i used per unit of activity j;

 p_i is the cost of resource i; and

n is the number of *j* activities.

subject to:
$$\sum_{j=1}^{n} a_{ij} \cdot x_{j} \le a_{i}$$
 $(i = 1, ..., m)$

The model attempts to maximise farm returns from irrigated agriculture in the light of land and water resource constraints and enterprise costs (part of which are directly associated with the cost of water) and returns. Consequently, the models are useful in looking at optimal responses to changes in variables such as water availability and price.

5. Assessment of environmental flow rules

NSW Agriculture was requested to evaluate a range of environmental flow management options developed by the LRMC. The initial request was to analyse five options and the results of this analysis were previously presented to the Committee and are discussed in section 5.3. The LRMC also requested the evaluation of a further two options (E73A and E131). The following discussion focuses on the evaluation of these two management options against base case conditions. The evaluation of the environmental flow rules is undertaken as an ex-ante analysis. A simulation approach is taken to evaluate the impact of flow rules over a range of climatic years from 1900/01 to 1996/97.

5.1 Evaluation of E73A and E131 environmental flow rules

The base case, E73A and E131 environmental flow rules are described in the form of hydrology data which specifics the availability of water to irrigators across different climatic years. Water availability is represented in the form of allocation announcements, which express the quantity of water available as a proportion of entitlement. For the purposes of this analysis, any access to off allocation flows is incorporated within the allocation data provided for each option. Simulated January announced allocations are used in this analysis, so any off allocation flows made available between August and January are reflected in the hydrology data (see Appendix 3).

The environmental flow rules E73A and E131 reflect specific cases of the flow rule categories outlined in Section 3 which included translucent releases, environmental contingency allowances and off allocation. One of the key differences between the environmental flow rules and the base case examined here is access to off allocation flows. Under the base case, off allocation access represents around 7 per cent of total water availability, while under the flow rules this falls to 3 and 0 per cent under E73A and E131, respectively. The hydrology data gave an average January announced allocation of 97.0 per cent under the base case and 73.1 and 68.3 per cent for E73A and E131 rules, respectively. The hydrology data also indicated that there was lower variability in terms of allocation availability for the two management options with lower standard deviations compared to the base case.

On the basis of the hydrology data provided, the agricultural impacts associated with the introduction of environmental flow rules are estimated across the six representative farms. The results of the analysis are reported in Table 3 and the impacts are shown graphically in Figure 6.

The results show that the introduction of LRMC's E73A environmental flow rules resulted in a decrease in whole farm gross margin of between 1 and 4 per cent and a decrease in net farm income of between 4 and 22 per cent across the six representative farms. In comparison, the introduction of E131 environmental flow rules resulted in a decrease in whole farm gross margin of between 2 and 6 per cent and a decrease in net farm income of between 5 and 32 per cent across all farms. Consequently, we can conclude that the introduction of E131 environmental flow rules marginally higher impacts on farms in the Lachlan catchment than E73A environmental flow rules.

	C 71A Bas	se Case		E 73.	A Rules			E 13	1 Rules		
	average Ja allocation	anuary 97.0 %	Ave	rage Januar	y allocation 73	.1 %	average January allocation 68.3%				
	Mean (\$)	SD (\$)	Mean (\$)	SD (\$)	Impact (\$)	Impact %	Mean (\$)	SD (\$)	Impact (\$)	Impact %	
Zone 1 Farm											
Gross Margin	88,021	12,753	85,039	14,462	-2,983	-3.4	83,779	15,349	-4,242	-4.8	
Net Farm Income	33,576	12,753	30,593	14,462	-2,983	-8.9	29,334	15,349	-4,242	-12.6	
Zone 2 Farm											
Gross Margin	159,854	15,281	154,336	14,407	-5,518	-3.5	152,736	15,113	-7,119	-4.5	
Net Farm Income	60,702	15,281	55,183	14,407	-5,518	-9.1	53,583	15,113	-7,119	-11.7	
Zone 3 Small Farm								_			
Gross Margin	110,921	8,369	107,669	6,891	-3,252	-2.9	106,577	7,439	-4,344	-3.9	
Net Farm Income	50,668	8,369	47,416	6,891	-3,252	-6.4	46,324	7,439	-4,344	-8.6	
Zone 3 Large Farm											
Gross Margin	230,119	9,985	226,873	9,983	-3,246	-1.4	225,788	10,944	-4,331	-1.9	
Net Farm Income	89,490	9,985	86,244	9,983	-3,246	-3.6	85,159	10,944	-4,331	-4.8	
Zone 4 Farm											
Gross Margin	431,923	25,232	422,974	28,344	-8,949	-2.1	420,166	31,785	-11,756	-2.7	
Net Farm Income	226,977	25,232	218,028	28,344	-8,949	-3.9	215,221	31,785	-11,756	-5.2	
Zone 5 Farm											
Gross Margin	149,233	23,748	142,787	27,730	-6,446	-4.3	140,068	30,189	-9,164	-6.1	
Net Farm Income	29,036	23,748	22,590	27,730	-6,446	-22.2	19,872	30,189	-9,164	-31.6	

Assessment of environmental flow rules

Table 3: Results of LRMC's E73A and E131 flow rules analysis

21



Figure 6: LRMC's E73A and E131 flow rules – Impact on Farm Gross Margin

Further analysis was undertaken to determine whether the agricultural impacts of environmental flows were statistically significant. The analysis found that the impacts of E131 environmental flow rules across the 97 year simulation period were found to be statistically significant at a 95 per cent confidence level for all the representative farms (all t-statistics above the critical value of 1.65). That means agricultural returns under situations of E131 environmental flows are consistently lower than without such flows for all farms. Impacts of E73A flow rules were found to be significant at a 95 per cent confidence level for five out of the six farms. The Zone 1 representative farm impacts became significant at 90 per cent confidence level. This is due to the relative high variation in farm returns for this representative farm.

The whole farm gross margin impacts from the LRMC rules (over the full range of climatic years) can be compared to the indicative flow rules developed by government agencies. NSW Agriculture conducted a regional analysis of the agricultural impacts on indicative flow rules in 1998. The impacts of indicative flow rules on regional gross margin were found to be between 4-5 per cent. The analysis of LRMC's E73A environmental flow rules shows lower impacts than the indicative flow rules for all the representative farms. The analysis of E131 environmental flow rules shows equal or lower impacts than the indicative flow rules for all but one of the representative farms (the farm in Zone 5 - Jemalong that has a high level of license activation). This suggests some progress in option development in reducing the negative effects of environmental flows on farm incomes.

The results also show that environmental flows are unlikely to have uniform impacts across farms in the catchment. This is due to a variety of farm specific factors including the current level of license activation, the profitability of existing land uses and the current financial position of the farm.

5.2 Impact of E73A and E131 environmental flow rules in Dry Years

The LRMC expressed interest in finding out the nature of agricultural impacts of environmental flow rules in dry years. This issue was addressed in this study by drawing on information already available from the main analysis which provides a time series of results for the base case and environmental flows over 97 climatic years. Given the availability of these results, essentially all that is required is a set of criteria to define dry years.

In some studies, dry years are defined through reference to historical rainfall conditions. A potential problem with this approach to irrigation issues in large inland catchments is the geographical separation of the main irrigated regions from the actual storages from which irrigation supplies are released. Using historical rainfall availability as an indicator of a dry year could indicate those years where dry seasonal conditions exist in the region but may also coincide with relatively abundant irrigation supplies. To overcome this problem, other approaches were considered.

A dry year from an irrigation perspective is probably best viewed in terms of the availability of irrigation supplies. A dry year was defined as any year, which reported a January allocation of less than 70 per cent for the base case. This definition yielded 26 dry years out of the total simulation period. The hydrology data alone illustrates that the proportional reductions to allocation resulting from environmental flow policies are larger in drier years relative to average years. The average January announced allocation is 35.0 per cent under the base case and 23.1 and 18.3 per cent respectively under E73A and E131 rules for the 26 dry years assessed. The impacts of E73A and E131 environmental flow rules in dry years are reported in Table 4.

The impact of E73A and E131 environmental flow rules was found to be much more significant in dry years. The impact of environmental flows on farm returns increased from between \$3,000-\$11,800 in an average year to between \$7,000-\$23,300 in a dry year. The representative farm in Zone 4 had access to groundwater and was the least affected in both absolute and percentage terms. As expected, the groundwater supplementation acted to offset the impact of reduced surface water access in dry years.

The higher impacts of environmental flows in dry years felt by most of the farms is a product of both a larger reduction in allocations during dry years and the higher marginal value of water derived from irrigated agriculture during periods of resource scarcity. The agricultural impacts of environmental flows are further increased in dry years because of the higher marginal value of water derived from irrigated agriculture during periods of resource scarcity. That is, the allocation of water away from irrigated agriculture will have a higher per unit cost during times of resource scarcity as higher returning enterprises are sacrificed from production. These impacts may have been even higher in the absence of measures implemented by the LRMC specifically to mitigate the effects of environmental flows in drier years.

	C 71A Base Case		E 73A Rules			E 131 Rules		
	average January allocation 35.0 %	average Ja	anuary allocation	23.1 %	average J	anuary allocation	n 18.3%	
	Mean (\$)	Mean (\$)	Impact (\$)	Impact %	Mean (\$)	Impact (\$)	Impact %	
Zone 1 Farm							•	
Gross Margin	73,824	66,780	-7,044	-9.5	63,589	-10,235	-13.9	
Net Farm Income	19,378	12,334	-7,044	-36.4	9,144	-10,235	-52.8	
Zone 2 Farm								
Gross Margin	147,866	136,376	-11,490	-7.8	132,339	-15,527	-10.5	
Net Farm Income	48,713	37,223	-11,490	-23.6	33,186	-15,527	-31.9	
Zone 3 Small Farm								
Gross Margin	105,419	99,947	-5,472	-5.2	97,069	-8,350	-7.9	
Net Farm Income	45,167	39,694	-5,472	-12.1	36,816	-8,350	-18.5	
Zone 3 Large Farm								
Gross Margin	220,993	215,167	-5,826	-2.6	212,471	-8,521	-3.9	
Net Farm Income	80,364	74,538	-5,826	-7.2	71,843	-8,521	-10.6	
Zone 4 Farm								
Gross Margin	406,688	393,937	-12,752	-3.1	386,226	-20,462	-5.0	
Net Farm Income	201,742	188,991	-12,752	-6.3	181,281	-20,462	-10.1	
Zone 5 Farm								
Gross Margin	120,299	104,303	-15,996	-13.3	97,018	-23,280	-19.4	
Net Farm Income	102	-15,894	-15,996	NA*	-23,178	-23,280	NA*	

Assessment of environmental flow rules

Table 4: Results of LRMC's E73A and E131 flow rules analysis in Dry Years

* Percentages are large because initial Net Farm Income for base case is very low. It is reasonable to expect some farms make a loss in dry years.

24

5.3 Assessment of previous options

NSW Agriculture was previously requested to evaluate five options and the results of this analysis are shown in Appendix 4. The hydrology data provided by DLWC (see Appendix 3) for this analysis showed only a marginal difference between the options in terms of average allocation availability. The average January allocations ranged from 68.3 to 69.6 per cent across the options. The hydrology data also indicated that there was no significant difference between the options in terms of allocation variability with standard deviations found to be the same across the options. Consequently, for the purposes of the analysis only the management options reporting the highest average value (E98) and the lowest average value (E116) were assessed.

It is important to note that the results of this analysis are not directly comparable with that of the main analysis. This is due to the change in the IQQM model as it was being periodically updated. As a consequence the hydrology received for the base case (C71) was different than for the current situation.

6. Summary

The extent of trade-offs associated with establishing environmental allocations is a significant issue in the Lachlan Catchment. There are important river health issues in the catchment as well as a large irrigation industry dependent upon secure irrigation supplies. The agricultural trade-offs incurred through the introduction of environmental flows requires analysis at two levels. First, at a broader regional scale, to determine the overall economic efficiency of options. Second, at a more disaggregated level to provide WMCs with distributional information on how subsets of the population might be affected. The on-farm impacts provide this distributional information and are the focus of this study.

A representative farm approach was used to assess the financial impacts on agriculture from the implementation of the environmental flows proposed by the LRMC. Six representative farms were developed for the analysis to reflect the typical farming systems in 5 principal production zones moving from east to west in the catchment. The impacts have been determined by quantifying the difference in farm returns between a base case (without environmental flows) and environmental flow scenarios. The evaluation of the impact of environmental flows was undertaken with climatic and hydrology simulation data to reflect the impacts over a range of climatic years from 1900/01 to 1996/97.

The results from the analysis show that the introduction of LRMC's E73A environmental flow rules resulted in a decrease in whole farm gross margin of between 1 and 4 per cent and a decrease in net farm income of between 4 and 22 per cent across the six representative farms. In comparison, the introduction of E131 environmental flow rules resulted in a decrease in whole farm gross margin of between 2 and 6 per cent and a decrease in net farm income of between 5 and 32 per cent across all farms. Consequently, we can conclude that the introduction of E131 environmental flow rules on farms in the Lachlan catchment than E73A environmental flow rules.

The agricultural returns under situations of E131 environmental flows are consistently lower than without such flows for all farms, while the same under E73A flow rules could be inferred for the case of most of the representative farms.

The whole farm gross margin impacts from the LRMC rules (over the full range of climatic years) were also compared to the indicative flow rules developed by government agencies (impact of indicative rules on average regional gross margin was previously found to be 4.2 per cent). This analysis showed that the rules developed by LRMC generated lower impacts for most of the farms.

The impact of E73A and E131 environmental flow rules was found to be much more significant in dry years. The impact of environmental flows on farm returns increased from between \$3,000-\$11,800 in an average year to between \$7,000-\$23,300 in a dry year. The higher impacts of environmental flows in dry years is a product of both a larger reduction in allocations during dry years and the higher marginal value of water derived from irrigated agriculture during periods of resource scarcity. These impacts may have been even higher in the absence of measures implemented by the LRMC specifically to mitigate the effects of environmental flows in drier years. The representative farms used in this study provide a representation of the predominant broadacre irrigation farming systems relying on surface water in the catchment. In this context, the study provides the LRMC with some indication as to how environmental flows may affect typical broadacre farms within the central irrigation areas of the Lachlan Valley. Obviously these representative farms do not represent the full diversity of farms in the region so some care should be exercised in generalising the results of this study. Ultimately, the impacts of environmental flows on any one farm will reflect a variety of farm specific factors including the current level of license activation, the productivity of existing land uses, the adjustment responses adopted to reduced water availability and the current financial position of the farm. Nevertheless, the results of this study should provide the LRMC with some indications on the likely impacts to be felt by broadcare farms in the catchment from the implementation of environmental flow policies.

7. References

Australian Bureau of Agricultural and Resource Economics, 1999. Structural Adjustment and Irrigated Broadacre Agriculture in the Southern Murray Darling Basin. Report to the Murray Darling Basin Commission, ABARE, Canberra.

Department of Land and Water Conservation, 1996. The Draft Lachlan Valley Strategic Water Management Plan 1996-2000. DLWC, Central West Region.

Department of Land and Water Conservation, 1998. *The Impact of Environmental Flow Rules* on the Regional Economy in the Lachlan Valley. A Report submitted to the Lachlan River Management Committee, Resource Economics Unit, DLWC.

Department of Land and Water Conservation, 2001. Preliminary Draft Water Sharing Plan for the Lachlan Regulated River. DLWC Central West Region.

Donovan, 1998. New South Wales Irrigation: Food and Fibre for a Secure, Caring and Sustainable World. NSW Irrigators' Council, North Revesby.

Environment Protection Authority, 1996. Proposed Interim Environmental Objectives for NSW Waters.

Jayasuriya, R., Catt, C. and Young, R. T., 1999. Local Consensus Data technique – applicability in the socio-economic analysis of water reforms. Discussion paper, Economic Services Unit, NSW Agriculture, Orange.

Lachlan Irrigation Research and Advisory Council Incorporated, 1997. Market audit of Lachlan Valley irrigated crop production. NSW Agriculture, Orange.

Lachlan Irrigation Research and Advisory Council Incorporated, 1999. Irrigated industries in the Lachlan Valley – future directions. LIRAC Inc., Forbes.

NSW Agriculture, 1996. An Economic Analysis of Interim River Flow Objective Scenarios in the Lachlan Valley. Final Report to the River Flows Objective Working Group, Economic Services Unit, NSW Agriculture, Orange.

NSW Agriculture, 1998. An Economic Analysis of Indicative Flow Rules in the Lachlan Valley. Report to the River Management Committee, Economic Services Unit, NSW Agriculture, Orange.

Snowy Water Inquiry, 1998. Snowy Water Inquiry Issues Paper. Snowy Water Inquiry, Governor Macquarie Tower, Sydney.

Appendix 1: Representative farm data

1.1 – Zone 1 Farm

					As in zone 2 t	out smaller fa	ms			
Farm areas										
Total farm area	Ha	304	range 150-600							
Area set up for irrigation	На	152								
Area normally irrigated	Ha	110	range 38-130							
Dryland area	На	194								
Beef cattle number	no.									
heep number	no.	700				Farm "plan"	showing rot	ation		
						Irrigated 152	ha	Dryland 152	ha	
later supplies						PP-L1		Canola - 19 I	ha	
egulated water						PP-L2		W		
irrigation entitlement	ML	600	range 150-100	D	475.0	IPP-L3	the second second	W	A CONTRACT	
access to off allocation	ML					PP-L4		PP-L		
nregulated water	11-				54.2	IPP-LS		PP-L		
licenced area	На				51.3	C		PP-N		
roundwater supplies					40.5	W AD L		PP-N		
Irrigation entitlement	NIL	Course SCC			ECC 0	Total water		PP-N		J
losest raimali site	location	or 2 ADS/A	irPort / PoctOf	fice	0.00C	Total water t	Ise			
arm labour		UTI. ANOTA		iice						
owner/manager	no, of weeks	50								
family	no. of weeks	50								
pernament labour	no. of weeks									
casual	no. of weeks	25								
the second se	and the second									
Irrigation characteristics										
		Area	Irrigation eff							
rigation by layout and method		Ha	%							
land formed - flood (LFF)		2007 Store 5								
land formed - spray (LFS)										
non landformed - flood (NLFF)										
non landformed - spray (NLFS)		152	80							
other										
igation infra-structure										
ver pump details										
capacity	MI/hour									
running cost	\$/hour									
ound water pump details										
capacity	Ml/hour									
running cost	\$/hour									
n-tarm storage										
sunace area	square metres									
depth	metres									
n-rarm recycling system	1/11									
cheme details										
name	name									
usage charge	\$/ML									
lixed charge				numping cost	fixed east					
			usage	pumping cost	fixed cost					
rater resource cost	EAA	47 20	2 00	42.40	2.07					
regulated supplies	\$/ML	47.20	3.80	43.40	3.07					
regulated supplies unregulated supplies	\$/ML \$/ML \$/MI	47.20	3.80	43.40	3.07					
arer resource cost regulated supplies unregulated supplies groundwater supplies	\$/ML \$/ML \$/ML	47.20	3.80	43.40	3.07					
ater resource cost regulated supplies groundwater supplies groundwater supplies ite : Water Use from model given belov	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont	3.80 hly rainfall da	43.40 ta and therefo	3.07 re all Gross M	argin figure	s here are b	ased on the	se average data	a.
ater resource cost regulated supplies groundwater supplies rte : Water Use from model given below	\$/ML \$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont	3.80 hly rainfall da	43.40 ta and therefo	3.07 re all Gross M	argin figure	s here are b	ased on the	se average data	a.
ater resource cost regulated supplies groundwater supplies nte : Water Use from model given below . Irrigated enterprises	\$/ML \$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area	3.80 hly rainfall dat expected water	43.40 ta and therefo water use from	3.07 re all Gross M Pump/Delivery	argin figure Yield	s here are b Price	ased on the V.Costs	se average data Gross Margin	a. Total GN
ater resource cost ergulated supplies groundwater supplies rte : Water Use from model given below . <i>Irrigated enterprises</i>	\$/ML \$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha	3.80 hly rainfall dat expected water use Ml/ha	43.40 ta and therefo water use from model Ml/ha	3.07 re all Gross M Pump/Delivery Cost \$/MI	a rgin figure Yield tonnes/ha	s here are b Price \$/tonne	ased on the V.Costs \$/ha	se average data Gross Margin \$/ha	a. Total GM
ater resource cost ergulated supplies groundwater supplies ote : Water Use from model given below . Irrigated enterprises inter crops	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha	3.80 hly rainfall dat expected watei use Ml/ha	43.40 ta and therefo water use from model MI/ha	3.07 re all Gross M Pump/Delivery Cost \$/MI	a rgin figure Yield tonnes/ha	s here are b Price \$/tonne	ased on the V.Costs \$/ha	se average dat Gross Margin \$∕ha	a. Total Gł
aterresource cost regulated supplies groundwater supplies tte : Water Use from model given below . Irrigated enterprises inter crops Short Fallow - Wheat	\$/ML \$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha	3.80 hly rainfall dat expected water use Ml/ha	43.40 ta and therefo water use from model MI/ha	3.07 re all Gross M Pump/Delivery Cost \$/MI	a rgin figure Yield tonnes/ha	s here are b Price \$/tonne	vased on the V.Costs \$/ha	se average dat Gross Margin \$/ha	a. Total GI
ater resource cost ergulated supplies groundwater supplies ste : Water Use from model given below . frrigated enterprises inter crops Short Fallow - Wheat Long Fallow - Wheat	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha 15	3.80 hly rainfall da expected water use Ml/ha 2.70	43.40 ta and therefo water use from model MI/ha 1.48	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20	argin figure Yield tonnes/ha	s here are b Price \$/tonne \$145.00	ased on the: V.Costs \$/ha \$383.22	se average data Gross Margin \$/ha \$341.78	a. Total Gf \$5,1
ater resource cost ergulated supplies groundwater supplies ote : Water Use from model given below . Irrigated enterprises inter crops Short Fallow - Wheat Long Fallow - Wheat Darley	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha 15	3.80 hly rainfall da expected water use MI/ha 2.70	43.40 ta and therefor water use from model MI/ha 1.48	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20	argin figure Yield tonnes/ha 5	s here are b Price \$/tonne \$145.00	ased on the V.Costs \$/ha \$383.22	se average dat: Gross Margin \$/ha \$341.78	a. Total Gf \$5,1
ater resource cost regulated supplies groundwater supplies sote : Water Use from model given below . Irrigated enterprises inter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Concle	\$/ML \$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha 15	3.80 hly rainfall dat expected water use MI/ha 2.70	43.40 ta and therefor water use from model MI/ha 1.48	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20	largin figure Yield tonnes/ha 5	s here are b Price \$/tonne \$145.00	v.Costs V.Costs \$/ha \$383.22	se average dat Gross Margin \$/ha \$341.78	a. Total Gi \$5,1
ater resource cost ergulated supplies groundwater supplies ste : Water Use from model given below . Irrigated enterprises inter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average montl Area Ha 15 19	3.80 hly rainfall dat expected water use MI/ha 2.70 2.70	43.40 ta and therefor water use from model MI/ha 1.48 1.50	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 5	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98	se average data Gross Margin \$/ha \$341.78 \$206.02	a. Total Gł \$5,1 \$3,5
ate resource cost ergulated supplies groundwater supplies ste : Water Use from model given below . Irrigated enterprises inter crops Short Fallow - Wheat Sont Fallow - Wheat Jarley Dats Sanola Dickpeas Enkoheane	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha 15 19	3.80 hly rainfall da expected water use MI/ha 2.70 2.70	43.40 ta and therefor water use from model MI/ha 1.48 1.50	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 5	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98	se average dat Gross Margin \$/ha \$341.78 \$206.02	a. Total Gł \$5,1 \$3,5
ater resource cost ergulated supplies groundwater supplies ste : Water Use from model given below . Irrigated enterprises inter crops Short Fallow - Wheat Long Fallow - Wheat Barley Dats Canola Chickpeas Fababeans	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha 15 19	3.80 hly rainfall da expected water use Ml/ha 2.70 2.70	43.40 ta and therefor water use from model Mi/ha 1.48 1.50	3.07 re all Gross M Pump/Delivery Cost \$//MI 47.20 47.20	Yield tonnes/ha 5	s here are b Price \$/tonne \$145.00 \$320.00	vased on the V.Costs \$/ha \$383.22 \$433.98	se average dat Gross Margin \$/ha \$341.78 \$206.02	a. Total Gł \$5,1 \$3,5
ater resource cost ergulated supplies groundwater supplies ste : Water Use from model given below . Irrigated enterprises inter crops Short Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha 15 19	3.80 hly rainfall da expected water use MI/ha 2.70 2.70	43.40 ta and therefo water use from model Mi/ha 1.48 1.50	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 5 2	s here are b Price \$/tonne \$145.00 \$320.00	vased on the V.Costs \$/ha \$383.22 \$433.98	se average dat: Gross Margin \$/ha \$341.78 \$206.02	a. Total GI \$5, \$3,
ate resource cost ergulated supplies groundwater supplies ste : Water Use from model given below . Irrigated enterprises inter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha 15 19	3.80 hly rainfall da expected water use MI/ha 2.70 2.70	43.40 ta and therefor water use from model MI/ha 1.48 1.50	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 5 2	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98	se average dat Gross Margin \$/ha \$341.78 \$206.02	a. Total GI \$5, \$3,5
ater resource cost ergulated supplies groundwater supplies ste : Water Use from model given below . Irrigated enterprises inter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins immer crops Cotton	\$/ML \$/ML \$/ML w is for 97 year a	47.20 Avea Ha 15 19	3.80 hly rainfall da expected water use MI/ha 2.70 2.70	43.40 ta and therefor water use from model MI/ha 1.48 1.50	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	Yield Yield tonnes/ha 5 2	shere are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98	se average dat Gross Margin \$/ha \$341.78 \$206.02	a. Total GI \$5, \$3,5
ater resource cost regulated supplies groundwater supplies groundwater supplies ote : Water Use from model given belor . Irrigated enterprises inter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins immer crops Cotton Sorahum	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha 15 19	3.80 hly rainfall da expected water use Ml/ha 2.70 2.70	43.40 ta and therefo water use from model MI/ha 1.48 1.50	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 5 2	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98	se average dat Gross Margin \$/ha \$341.78 \$206.02	a. Total GI \$5, \$3,9
ater resource cost ergulated supplies groundwater supplies ste : Water Use from model given belor . Irrigated enterprises inter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins immer crops Cotton Sorghum Rice	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha 15 19	3.80 hly rainfall da expected water use MI/ha 2.70 2.70	43.40 ta and therefor water use from model Mi/ha 1.48 1.50	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figurea Yield tonnes/ha 5 2	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98	se average dat Gross Margin \$/ha \$341.78 \$206.02	a. Total Gł \$5,1 \$3,5
ater resource cost ergulated supplies groundwater supplies ste : Water Use from model given belor . frrigated enterprises inter crops Short Fallow - Wheat Long Fallow - Wheat Barley Dats Canola Chickpeas Fababeans Lupins Ummer crops Cotton Sorghum Rice Soybeans	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha 15 19	3.80 hly rainfall da expected water use MI/ha 2.70 2.70	43.40 ta and therefor water use from model MI/ha 1.48 1.50	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	Yield Yield tonnes/ha 5 2	shere are b Price \$/tonne \$145.00 \$320.00	ased on the V. Costs \$/ha \$383.22 \$433.98	se average dat Gross Margin \$/ha \$341.78 \$206.02	a. Total Gł \$5,1 \$3,5
ate resource cost eregulated supplies groundwater supplies ste : Water Use from model given belor . Irrigated enterprises inter crops Short Fallow - Wheat Sarley Dats Sarley Dats Sarley Dats Sarbabeans -Jupins mmer crops Dotton Sorghum Rice Soybeans Maize	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha 15 19	3.80 hly rainfall da expected water use Ml/ha 2.70 2.70	43.40 ta and therefor water use from model MI/ha 1.48 1.50	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 5 2	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98	se average dat Gross Margin \$/ha \$341.78 \$206.02	a. Total Gł \$5,1 \$3,5
ate resource cost are resource cost are regulated supplies aroundwater supplies ste : Water Use from model given belor . Irrigated enterprises inter crops Short Fallow - Wheat Barley Dats Canola Chickpeas "ababeans upins mmer crops Cotton Sorghum Nice Soybeans Maize Sunfowers	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha 15 19	3.80 hly rainfall da expected water use MI/ha 2.70 2.70	43.40 ta and therefor water use from model Mi/ha 1.48 1.50	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 5 2	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98	se average dat Gross Margin \$/ha \$341.78 \$206.02	a. Total GI \$5, \$3,
ate resource cost ergulated supplies groundwater supplies inte : Water Use from model given belor . Irrigated enterprises inter crops Short Fallow - Wheat Long Fallow - Wheat Darts Canola Chickpeas Fababeans Lupins mmer crops Cotton Sorghum Rice Soybeans Maize Sunflowers	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha 15 19	3.80 hly rainfall da expected water use MI/ha 2.70 2.70	43.40 ta and therefor water use from model MI/ha 1.48 1.50	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figurea Yield tonnes/ha 5 2	s here are b Price \$/tonne \$145.00 \$320.00	ased on the ∨ Costs \$/ha \$383.22 \$433.98	se average dat Gross Margin \$/ha \$341.78 \$206.02	a. Total GI \$5, \$3,5
ate resource cost ergulated supplies groundwater supplies ste : Water Use from model given belor . frrigated enterprises Short Fallow - Wheat Cong Fallow - Wheat Barley Dats Sanela Chickpeas Fababeans - uppins mmer crops Cotton Sorghum Rice Sorghans Maize Sunflowers sture	\$/ML \$/ML \$/ML w is for 97 year a (LSM's/ha or an	47.20 average mont Area Ha 15 19 indication of st	3.80 hly rainfall da expected water use Ml/ha 2.70 2.70 2.70	43.40 water use from model Mi/ha 1.48 1.50	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 5 2	s here are b Price \$/tonne \$145.00 \$320.00	vased on the V.Costs \$/ha \$383.22 \$433.98	se average dat Gross Margin \$/ha \$341.78 \$206.02	a. Total GI \$5, \$3,
ate resource cost ergulated supplies groundwater supplies ste : Water Use from model given befor . Irrigated enterprises inter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas afababeans Lupins Immer crops Cotton Sorghum Rice Sorgheans Maize Sunflowers Sufformers Sture Luceme hay	\$/ML \$/ML \$/ML w is for 97 year a (LSM's/ha or an	47.20 average mont Area Ha 15 19 indication of st 76	3.80 hly rainfall da expected water use MI/ha 2.70 2.70 2.70	43.40 ta and therefor water use from model MI/ha 1.48 1.50 5.53	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 5 2 2 12.50	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98 \$433.98	se average dat Gross Margin \$/ha \$341.78 \$206.02 \$206.02	a. Total GI \$5, \$3, \$3, \$43,0
ate resource cost ergulated supplies groundwater supplies ste : Water Use from model given belor . Irrigated enterprises inter crops Short Fallow - Wheat Daney Dats Canola Chickpeas Fababeans Lupins Immer crops Cotton Sorghum Rice Soybeans Maize Sunflowers sture Lucerne hay Lucerne hay Lucerne hay Lucerne hay	\$/ML \$/ML \$/ML w is for 97 year of (LSM's/ha or an	47.20 average month Area Ha 15 19 indication of st 76	3.80 hly rainfall da expected water use MU/ha 2.70 2.70 2.70 5.25	43.40 ta and therefor model Ml/ha 1.48 1.50	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 5 2 2 12.50	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98 \$433.98 \$1,184.18	se average dat Gross Margin \$/ha \$341.78 \$206.02 \$265.82	a. Total GI \$5, \$3,5 \$3,5 \$3,5 \$3,5 \$43,6
ater resource cost regulated supplies groundwater supplies groundwater supplies sote : Water Use from model given belor . firigated enterprises inter crops Short Fallow - Wheat Dats Canola Chickpeas Fababeans Lupins immer crops Cotton Sorghum Rice Soybeans Valaze Sunflowers sture Luceme hay Lummer pasture (perennial based) - Spring	\$/ML \$/ML \$/ML w is for 97 year a (LSM's/ha or an	47.20 average mont Area Ha 15 19 indication of st 76	3.80 hly rainfall da expected water use Ml/ha 2.70 2.70 2.70	43.40 ta and therefor water use from model MI/ha 1.48 1.50 5.53	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 2 2 12.50	shere are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98 \$433.98	se average dat Gross Margin \$/ha \$341.78 \$206.02 \$266.02	a. Total Gl \$5, \$3, \$3,
ate resource cost ergulated supplies groundwater supplies ste : Water Use from model given befor . Irrigated enterprises inter crops Short Fallow - Wheat Barley Oats Canola Chickpeas ababeans Lupins immer crops Cotton Sorghum Rice Sorgheans Maize Sunflowers sture Luceme hay Jummer pasture (perennial based) - Spring - Summer	\$/ML \$/ML \$/ML w is for 97 year a (LSM's/ha or an	47.20 average mont Area Ha 15 19 indication of st	3.80 hly rainfall dat expected water use MI/ha 2.70 2.70 2.70	43.40 ta and therefor water use from nodel Mi/ha 1.48 1.50 5.53	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 5 2 2 12.50	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98 \$433.98	se average dat Gross Margin \$/ha \$341.78 \$206.02 \$565.82	a. Total Gi \$5, \$3, \$3, \$43,0
ater resource cost ergulated supplies groundwater supplies ste : Water Use from model given belor . Irrigated enterprises inter crops Short Fallow - Wheat Long Fallow - Wheat Canola Chickpeas Fababeans Lupins immer crops Cotton Sorghum Rice Soybeans Maize Sunflowers isture Luceme hay Lummer pasture (perennial based) - Spring - Summer - Summer	\$/ML \$/ML \$/ML w is for 97 year a	47.20 average mont Area Ha 15 19 indication of st 76	3.80 hly rainfall da expected water use MI/ha 2.70 2.70 2.70	43.40 ta and therefor water use from model MI/ha 1.48 1.50 5.53	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 2 2 12.50	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98 \$433.98 \$1,184.18	se average dat Gross Margin \$/ha \$341.78 \$206.02 \$265.82	a. Total GI \$5, \$3,5 \$3,5 \$3,5 \$3,5 \$43,6
ater resource cost regulated supplies groundwater supplies ste : Water Use from model given belor . frrigated enterprises inter crops Short Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins Tababeans Lupins Tababeans Soghans Maize Soghans Maize Sunflowers Isture Luceme hay Lummer pasture (perennial based) - Spring - Summer - Autumn	\$/ML \$/ML \$/ML w is for 97 year a (LSM's/ha or an	47.20 average mont Area Ha 15 19 indication of st 76	3.80 hly rainfall da expected water use Ml/ha 2.70 2.70 2.70	43.40 ta and therefor water use from model MI/ha 1.48 1.50 5.53	3.07 re all Gross M Pump/Delivery Cost \$//MI 47.20 47.20	argin figure Yield tonnes/ha 5 2 2 12.50	shere are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98 \$433.98 \$1,184.18	se average dat Gross Margin \$/ha \$341.78 \$206.02 \$266.02	a. Total GI \$5, \$3, \$43,(
aler resource cost regulated supplies groundwater supplies groundwater supplies sote : Water Use from model given befor . Irrigated enterprises inter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins Immer crops Cotton Sorghum Rice Soybeans Maize Sunflowers Inter Luceme hay Iummer pasture (perennial based) - Spring - Summer - Autumn Winter Vinter pasture (subclover based) - Device	S/ML S/ML S/ML w is for 97 year a	47.20 average mont Area Ha 15 19 indication of st	3.80 hly rainfall da expected water use MI/ha 2.70 2.70 2.70	43.40 ta and therefor water use from nodel Mi/ha 1.48 1.50 5.53	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 5 2 2	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98 \$1,184.18	se average dat Gross Margin \$/ha \$341.78 \$206.02 \$565.82	a. Total Gł \$5,1 \$3,5 \$43,0
ate resource cost ergulated supplies groundwater supplies ste : Water Use from model given belor . Irrigated enterprises inter crops Short Fallow - Wheat Cong Fallow - Wheat Canola Chickpeas Fababeans Lupins immer crops Cotton Sorghum Rice Soybeans Maize Sunflowers sture 	S/ML S/ML S/ML w is for 97 year a	47.20 average mont Area Ha 15 19 indication of st 76	3.80 hly rainfall da expected water use MU/ha 2.70 2.70 2.70	43.40 ta and therefor model MI/ha 1.48 1.50 5.53	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 5 2 2	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98 \$1,184.18	se average dat Gross Margin \$/ha \$341.78 \$206.02 \$565.82	a. Total GI \$5, \$3,5 \$3,5 \$3,5 \$3,5 \$43,6
ate resource cost are resource cost are gulated supplies groundwater supplies ste : Water Use from model given belor . Irrigated enterprises inter crops Short Fallow - Wheat andey Dats andey Dats analey Dats Canola Chickpeas Fababeans .upins mmer crops Cotton Sorghum Rice Soybeans daize Sunflowers sture .uceme hay ummer pasture (perennial based) - Spring Summer .dutumn .Winter .inter pasture (subclover based) .Spring .Summer	\$/ML \$/ML \$/ML w is for 97 year a (LSM's/ha or an	47.20 average mont Area Ha 15 19 indication of st 76	3.80 hly rainfall da expected water use Ml/ha 2.70 2.70 2.70	43.40 ta and therefor water use from model MI/ha 1.48 1.50 5.53	3.07 re all Gross M Pump/Delivery Cost \$//MI 47.20 47.20	argin figure Yield tonnes/ha 2 2 12.50	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98 \$1,184.18	se average dat Gross Margin \$/ha \$341.78 \$206.02 \$565.82	a. Total Gl \$5, \$3, \$43,(
ate resource cost equilated supplies proundwater supplies integuilated supplies inter crops short Fallow - Wheat Barley Jats Canola Chickpeas a ababeans uppins inter crops Coston Sorghum Nice Sorghum Summer Autumn Minter Summer Autumn Minter Summer Autumn	S/ML S/ML S/ML w is for 97 year a	47.20 average mont Area Ha 15 19 indication of st 76	3.80 hly rainfall dat expected water use MI/ha 2.70 2.70 2.70	43.40 ta and therefor water use from model MI/ha 1.48 1.50 5.53	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 2 2	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98 \$1,184.18	se average dat Gross Margin \$/ha \$341.78 \$206.02 \$565.82	a. Total G \$5, \$3; \$3;
ate resource cost equilated supplies groundwater supplies te : Water Use from model given belor 	S/ML S/ML S/ML w is for 97 year a	47.20 average mont Area Ha 15 19 indication of st 76	3.80 hly rainfall da expected water use MI/ha 2.70 2.70 2.70	43.40 ta and therefor model MI/ha 1.48 1.50 5.53	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 5 2 2	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98 \$1,184.18	se average dat Gross Margin \$/ha \$341.78 \$206.02 \$565.82	a. Total Gi \$5, \$3; \$43,
ate resource cost equilated supplies groundwater supplies ite : Water Use from model given belor 	S/ML S/ML S/ML w is for 97 year a (LSM's/ha or an	47.20 average mont Area Ha 15 19 indication of st 76	3.80 hly rainfall da expected water use Ml/ha 2.70 2.70 2.70	43.40 ta and therefor water use from model MI/ha 1.48 1.50 5.53	3.07 re all Gross M Pump/Delivery Cost \$//MI 47.20 47.20	argin figure Yield tonnes/ha 2 2	s here are b Price \$/tonne \$145.00 \$320.00	sased on the V.Costs \$/ha \$383.22 \$433.98 \$1,184.18	se average dat Gross Margin \$/ha \$341.78 \$206.02 \$565.82	a. Total GI \$5, \$3, \$43,(
ate resource cost equilated supplies groundwater supplies te : Water Use from model given befor inrigated enterprises inter crops short Fallow - Wheat Barley Barley Dats Janela Anola Shickpeas Jababeans Jupins mmer crops Cotton Jorghum lice Joybeans daize Lunflowers sture ucerne hay unmer pasture (perennial based) Spring Summer Autumn Winter Inter pasture (subclover based) Spring Summer Autumn Winter Muther Cops Summer Autumn Winter Muther Cops Summer Autumn Winter Muther Cops Summer Autumn Winter Summer	S/ML S/ML S/ML w is for 97 year a	47.20 average mont Area Ha 15 19 indication of st 76	3.80 hly rainfall dat expected water use MI/ha 2.70 2.70 2.70 bocking rate) 6.25	43.40 ta and therefor model Ml/ha 1.48 1.50 5.53 5.53	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 2 2	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98 \$1,184.18	se average dat Gross Margin \$/ha \$341.78 \$206.02 \$565.82	a. Total GI \$5; \$3; \$43;0
ate resource cost equilated supplies groundwater supplies for the supplies ste : Water Use from model given belor 	S/ML S/ML S/ML w is for 97 year a	47.20 average mont Area Ha 15 19 indication of st 76	3.80 hly rainfall dar expected water use MU/ha 2.70 2.70 2.70 6.25	43.40 ta and therefor model MI/ha 1.48 1.50 5.53 471.2 ML	3.07 re all Gross M Pump/Delivery Cost \$/MI 47.20 47.20	argin figure Yield tonnes/ha 5 2 12.50	s here are b Price \$/tonne \$145.00 \$320.00	ased on the V.Costs \$/ha \$383.22 \$433.98	se average dat Gross Margin \$/ha \$341.78 \$206.02 \$565.82	a. Total G! \$3,5 \$3,5 \$3,5 \$43,0

Appendices

1.1 – Zone 1 Farm *Continued*

4. Dryland enterprises											
Winter crops							Area	Yield	V.Costs	Gross Margin	Total GM
- Short Fallow - Wheat							61	2.50	\$165.38	\$197.12	\$ \$12.02
- Long Fallow - wheat - Barley							01	2.50	4 105.50	157.12	\$12,02 \$
- Oats - Canola							19	1.50	\$350.14	\$129.86	\$ \$2,46
- Chickpeas											\$
- Fababeans - Lupins											5 5
Summer crops											9 9
- Cotton											9
- Sorghum - Sovbeans											ş
- Maize											9
- Sunnowers											ş
Pasture	(LS	SM's/ha or an	indication of s	stocking rate)			19	4.00	\$165.04	\$314.96	\$ \$5,98
- Improved pasture							95	link Sheep2>	\$209.74	\$213.78	\$20,30
- Spring - Summer											\$
- Autumn											5
- Unimproved pasture (native)											\$
- Spring - Summer											5
- Autumn											\$
- Other crops											\$
Dodand Gross Margin							194	На			\$40,78
							104				
E. Questioned as at atrusture											
5. Overnead cost structure											
Administration expenses		1400									
- accounting - bank charges		1800									
- insurance (farm & Vehicles)		2100	15%		ASSETS			Ane (urrent Value	Expected Life	
- telephone		800	of wages		AUDETO	and the second second second second					
- stationary Labour (permanent & casual)		14000			PLANT :				1000000		
Fuel and oil (farm vehicles)		2800			Tractor - 150 h	np			45000		
Repairs and maintenance		1200			Tractor - 75 h	יף ס			15000		
 plant and equipment structures 		2800 1200			Tractor - MF 3 Tractor with L	15 oader - 50 hp			6000		
Depreciation		17504	RATE		Header				22000		
 plant and equipment structures 		17524	6.8%		Rake	lioner			5000		
Rates - Stocking charges (PPB)		300			Rake				3000		
- Land		1230			Baler				15000		
- Water Other operating overheads (Rego & License)		2800			Sprayer Bale Loader o	r Wagon			20000		
Total operating overheads	\$	54,445.6			Augers Field Bins				8000		
					Plough				5000		
6. Profit and Financial analys	sis				Scarifier				2200		
OTHER FARM INCOME (eg timber)	5	-			Harrows - Two				5000		
TOTAL FARM GROSS MARGIN	s	92 828			Motor Bike				2500 15000		
	1	01,010			Ute				50000		
TOTAL OPERATING OVERHEADS	\$	54,446			Travelling Irrig	ator Plant			50000		
NET FARM INCOME	5	38,383			STRUCTURES	5			25000		
	1	24,000			Guildines				257700		
OPERATING RETURN (A)	5	14,383 F	component		Livestock Rams -	Number 20	Value/hd \$100	2000			
- OD interest paid		\$1,050	ro 171		Ewes -	700	\$30	21000			
- Term Loan interest		\$12,750	\$5,313		Cows -		\$450	0			
BUSINESS BETLIBN (B)	5	(4 167)			Liquid assets	such as					
		(11-1)			Bank Deposits	s - Off farm inve	stment	25,000	super		
- liquid assets	\$	35,000			Snares / Equi	,		10000			
- value of land and improvements	5	1,000,000			LIARII ITIES				Rate	Term	
- value of sheep	5	23,000			OD Bank			30000	10.5%	1	
- value of cattle Sub Total	5	1,315,700			HP / Lease Mortgage			50000 150000	9.5% 8.5%	5 15	
TOTAL LIABILITIES	ę	220.000				Bank lending	equity	82%			
IV THE LIMUILITIES	3	230,000					- 40113	02 78			
EQUITY (D)	\$	1,085,700									
EQUITY RATIO D/C X 100		82.5%									
RETURN ON TOTAL ASSETS (A) / (C) × 100 RETURN ON EQUITY (B) / (D) × 100		1.1% -0.4%									
OFF-FARM INCOME Approx. Taxable Farm Profit or Loss NET CASH RESULT (after tax)	\$ \$ 5	15,000 19,833 34,805									
DECISION TREES .											
DECISION TREES										i	

1.2 – Zone 2 Farm

Lucerne / Grazing / Winter	Crone 150 Eau									
	ciups · isurai	mers			As in zone	l but larger farms				
1 Dhusical farm characte	ristics									
1. Physical lanni character	115005									
Farm areas										
Total farm area	Ha	800								
Area set up for irrigation	Ha	320								
Area normally irrigated	Ha	160	range 160-180	1						
Dryland area	Ha	640								
Beef cattle number	no.	50	45 Vealers							
Sheep number	no.	1500	Lambs 1500							
						<u>Fi</u>	arm "plan" :	showing rot	ation	
Water supplies							igated 160	setup 160h	aUryland 480	ha
Regulated water	NI	1000		0		6U.UA	P-5	AP-S	001 200 5	
- imgation entitlement	ML	1000	range 600-180	U		90.083	P-L2	PP-LI	JPP-L 200 n	d
- access to on allocation	IVIL					000.0 F	P.1.4	1		
licenced area	На					P	P.15			
- incenced area Groupdwater supplies						84 0 0	2		Word 32) ha
- irrigation entitlement	MI					108.0	1			
Closest rainfall site	location	Forbes-Camp	St			W	/ - 20 ha	1		
		or ?: Bethany	Park/Airport A	ws/Muddy Wat	er/3	932.0 To	otal water u	se		
arm labour										
- owner/manager	no. of weeks	50								
family	no. of weeks									
pernament labour	no. of weeks	48								
casual	no. of weeks	25								
2. Irrigation characteristic	S									
		Area	Irrigation eff							
rigation by layout and method		Ha	%							
land formed - flood (LFF)		160	75							
land formed - spray (LFS)										
non landformed - 1000 (NLFF)										
non randiormed - spray (NLFS)										
UIICI										
rigation infra-structure										
iver pump details										
- capacity	Ml/hour									
- running cost	\$/hour									
round water pump details										
- capacity	MI/hour									
- running cost	\$/hour									
)n-farm storage										
- surface area	square metres									
- depth	metres									
On-farm recycling system	Y/N									
On-farm recycling system Scheme details	Y/N									
Dn-farm recycling system Scheme details name	Y/N name									
Dn-farm recycling system Icheme details name usage charge fixed charge	Y/N name \$/ML									
In-farm recycling system Icheme details name usage charge fixed charge	Y/N name \$/ML \$		anc su	numping cost	fixed cost					
Dn-farm recycling system Scheme details - name - usage charge - fixed charge Yater resource cost - renulated supplies	Y/N name \$/ML \$ \$/MI	14.96	usage 3 80	pumping cost 11.16	fixed cost 3.07					
Dn-farm recycling system Ccheme details name usage charge fixed charge Vater resource cost regulated supplies unreculated supplies	Y/N name \$/ML \$ \$/ML \$/ML	14.96	usage 3.80	pumping cost 11.16	fixed cost 3.07					
h-farm recycling system cheme details name usage charge fixed charge Vater resource cost regulated supplies unregulated supplies groundwater supplies	Y/N name \$/ML \$ \$/ML \$/ML \$/ML	14.96	usage 3.80	pumping cost 11.16	fixed cost 3.07					
h-farm recycling system cheme details name usage charge fixed charge Vater resource cost regulated supplies unregulated supplies groundwater supplies	Y/N name \$/ML \$ \$/ML \$/ML \$/ML	14.96	usage 3.80	pumping cost 11.16	fixed cost 3.07					
In-farm recycling system cheme details name usage charge fixed charge Vater resource cost regulated supplies unregulated supplies groundwater supplies ote : Water Use from model given be	Y/N name \$/ML \$ \$/ML \$/ML \$/ML \$/ML	14.96 age monthly ra	usage 3.80 sinfall data ar	pumping cost 11.16 Id therefore al	fixed cost 3.07	in figures here a	nre based (on these av	verage data.	
In-farm recycling system cheme details name usage charge fixed charge yater resource cost regulated supplies groundwater supplies groundwater supplies ote : Water Use from model given be a trrinated enformises	Y/N name \$/ML \$ \$/ML \$/ML \$/ML \$/ML	14.96 age monthly ra	usage 3.80 sinfall data ar	pumping cost 11.16 Id therefore al	fixed cost 3.07 I Gross Marg	in figures here a	ire based (on these av	verage data.	Total GM
In-farm recycling system cheme details name usage charge fixed charge Vater resource cost regulated supplies unregulated supplies groundwater supplies ote : Water Use from model given be B. Irrigated enterprises	Y/N name \$/ML \$ \$/ML \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area	usage 3.80 sinfall data ar expected wate	pumping cost 11.16 ad therefore al water use from	fixed cost 3.07 I Gross Marg Pump/Deliver	in figures here a Yield Pr	ire based (on these av V.Costs	verage data. Gross Margir	Total GM
n-farm recycling system cheme details name usage charge fixed charge fater resource cost regulated supplies unregulated supplies groundwater supplies ote : Water Use from model given be c. frrigated enterprises	Y/N name \$/ML \$ \$/ML \$/ML \$/ML \$/ML	14.96 age monthly ra Area Ha	usage 3.80 sinfall data ar expected wate use Ml/ha	pumping cost 11.16 Id therefore al water use from model MUha	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI	in figures here a 'Yield Pr tonnes/ha \$//	ire based d ice lonne	on these av V.Costs \$/ha	verage data. Gross Margir \$/ha	Total GM
n-farm recycling system cheme details name usage charge fixed charge (afer resource cost regulated supplies unregulated supplies groundwater supplies ote : Water Use from model given be c. Irrigated enterprises	Y/N name \$/ML \$ \$/ML \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha	usage 3.80 sinfall data ar expected wate use MI/ha	pumping cost 11.16 Id therefore al water use from model MVha	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI	in figures here a Yield Pr tonnes/ha \$/	re based o ice tonne	on these av V.Costs \$/ha	verage data. Gross Margir \$/ ha	Total GN
n-farm recycling system cheme details name usage charge fixed charge fixed charge valer resource cost regulated supplies groundwater supplies ote : Water Use from model given be c. irrigated enterprises Finter crops Short Fallow - Wheat	Y/N name \$/ML \$ \$/ML \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha	usage 3.80 sinfall data ar expected wate use MI/ha	pumping cost 11.16 ad therefore al water use from model Mi/ha	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI	in figures here a ∙Yield Pr tonnes/ha \$/1	ire based o	on these av V Costs \$/ha	verage data. Gross Margir \$/ha	Total GN
n-farm recycling system cheme details name usage charge fixed charge fixed charge fixed charge after resource cost regulated supplies unregulated supplies groundwater supplies ote : Water Use from model given be b; firtigated enterprises Finter crops Shoft Fallow - Wheat Long Fallow - Wheat Boden	Y/N name \$/ML \$ \$/ML \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40	usage 3.80 ainfall data ar expected wate use MI/ha 2.7	pumping cost 11.16 nd therefore al water use from model Mi/ha 1.70	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96	in figures here a Yield Pr tonnes/ha \$/h	ire based o ice iconne \$145.00	on these av V Costs \$/ha \$338.85	verage data. Gross Margir \$/ha \$386.15	Total GM \$15,4
n-farm recycling system cheme details name usage charge fixed charge fixed charge offer resource cost regulated supplies unregulated supplies groundwater supplies ote : Water Use from model given be b. frrigated enterprises Finter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oate	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40	usage 3.80 sinfall data ar expected wate use MI/ha 2.7	pumping cost 11.16 nd therefore al water use from model Ml/ha 1.70	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96	in figures here a Yield Pr tonnes/ha \$/1	ire based o ice tonne \$145.00	on these av ∨ Costs \$/ha \$338.85	verage data. Gross Margir \$/ha \$386.15	Total GN \$15,4
In-farm recycling system cheme details name usage charge fixed charge Vater resource cost regulated supplies unregulated supplies groundwater supplies ote : Water Use from model given be B. Irrigated enterprises Finter crops Short Fallow - Wheat Barley Oats Canola	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40	usage 3.80 sinfall data ar expected wate use MI/ha 2.7	pumping cost 11.16 ad therefore al water use from model Mi/ha 1.70	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96	in figures here a Yield Pr tonnes/ha \$/1 5	re based o ice tonne \$145.00 \$320.00	on these av V Costs \$/ha \$338.85 \$387.78	verage data. Gross Margir \$/ha \$386.15 \$252.22	Total GN \$15,4
n-farm recycling system cheme details name usage charge fixed charge fixed charge dater resource cost regulated supplies groundwater supplies ote : Water Use from model given be b: Irrigated enterprises Finter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickeess	Y/N name \$/ML \$ \$/ML \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40 20	usage 3.80 ainfall data ar expected wate use MI/ha 2.7 4.2	pumping cost 11.16 ad therefore al water use from model MI/ha 1.70 1.63	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2	rre based i ice tonne \$145.00 \$320.00	on these av V.Costs \$/ha \$338.85 \$387.78	verage data. Gross Margir \$/ha \$386.15 \$252.22	Totəl GN \$15,4 \$5,0
n-farm recycling system cheme details name usage charge fixed charge fixed charge diater resource cost regulated supplies groundwater supplies ote : Water Use from model given be c. Irrigated enterprises Finter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fibabeens	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40 20	usage 3.80 ainfall data ar expected wate use MI/ha 2.7 4.2	pumping cost 11.16 nd therefore al water use from model MI/ha 1.70 1.63	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/Mil 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2	rre based d ice tonne \$145.00 \$320.00	on these av V Costs \$/ha \$338.85 \$387.78	verage data. Gross Margir \$/ha \$386.15 \$252.22	Totəl GN \$15,4 \$5,0
n-farm recycling system cheme details name usage charge fixed charge fater resource cost regulated supplies unregulated supplies groundwater supplies ote : Water Use from model given be c. firtigated enterprises finter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40 20	usage 3.80 sinfall data ar expected wate use MI/ha 2.7 4.2	pumping cost 11.16 nd therefore al water use from model MI/ha 1.70 1.63	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5	ire based i ice tonne \$145.00 \$320.00	on these av V Costs \$/ha \$338.85 \$387.78	verage data. Gross Margir \$/ha \$386.15 \$252.22	Total GM \$15,4 \$5,0
In-farm recycling system Cheme details name usage charge fixed charge Vater resource cost regulated supplies unregulated supplies groundwater supplies Inter crops Short Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40 20	usage 3.80 sinfall data ar expected wate use MI/ha 2.7 4.2	pumping cost 11.16 Ind therefore al water use from model Mi/ha 1.70 1.63	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2	stre based of ice tonne \$145.00 \$320.00	on these av ∨ Costs \$/ha \$338.85 \$387.78	verage data. Gross Margir \$/ha \$386.15 \$252.22	Total GM \$15,4 \$5,0
h-farm recycling system Scheme details name usage charge fixed charge Valer resource cost regulated supplies groundwater supplies ote : Water Use from model given be B. Irrigated enterprises Vinter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40 20	usage 3.80 sinfall data ar expected wate use MVha 2.7 4.2	pumping cost 11.16 ad therefore al water use from model MI/ha 1.70 1.63	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2	stre based of ice forme \$145.00 \$320.00	on these av V.Costs \$/ha \$338.85 \$387.78	verage data. Gross Margir \$/ha \$386.15 \$252.22	Total GN \$15,4 \$5,0
In-farm recycling system cheme details name usage charge fixed charge fixed charge valer resource cost regulated supplies groundwater supplies ote : Water Use from model given be B. Irrigated enterprises Finter crops Short Fallow - Wheat Long Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40 20	usage 3.80 ainfall data ar expected wate use MI/ha 2.7 4.2	pumping cost 11.16 nd therefore al water use from model MI/ha 1.70 1.63	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/Mil 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2	sre based o ice tonne \$145.00 \$320.00	on these av V Costs \$/ha \$338.85 \$387.78	verage data. Gross Margir \$/ha \$386.15 \$252.22	Totəl GM \$15,4 \$5,0
h-farm recycling system cheme details name usage charge fixed charge Vater resource cost regulated supplies groundwater supplies lote : Water Use from model given be B. Irrigated enterprises Vinter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40 20	usage 3.80 sinfall data ar expected wate use MI/ha 2.7 4.2	pumping cost 11.16 ad therefore al water use from model MVha 1.70 1.63	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/h 5 2	stre based of tonne \$145.00 \$320.00	on these av ∨ Costs \$/ha \$338.85 \$387.78	verage data. Gross Margir \$/ha \$386.15 \$252.22	Total GN \$15,4 \$5,0
In-farm recycling system cheme details name usage charge fixed charge valer resource cost regulated supplies groundwater supplies ote : Water Use from model given be B. Irrigated enterprises Finter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40 20	usage 3.80 sinfall data ar expected wate use MI/ha 2.7 4.2	pumping cost 11.16 ad therefore al water use from model MVha 1.70 1.63	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2	stre based of tonne \$145.00 \$320.00	on these av ∨ Costs \$/ha \$338.85 \$387.78	verage data. Gross Margir \$/ha \$386.15 \$252.22	Total GM \$15,4 \$5,0
In-farm recycling system cheme details name usage charge fixed charge fixed charge valer resource cost regulated supplies groundwater supplies ote : Water Use from model given be B. Irrigated enterprises Finter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Soybeans	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40 20	usage 3.80 ainfall data ar expected wate use MI/ha 2.7 4.2	pumping cost 11.16 ad therefore al water use from model MI/ha 1.70 1.63	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2	stre based of ice iconne \$145.00 \$320.00	on these av ∨ Costs \$/ha \$338.85 \$387.78	verage data. Gross Margir \$/ha \$386.15 \$252.22	Total GM \$15,4 \$5,0
n-farm recycling system cheme details name usage charge fixed charge fixed charge fixed charge fixed supplies unregulated supplies groundwater supplies ote : Water Use from model given be c. firtigated enterprises finter crops Short Fallow - Wheat Long Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Soybeans Maize	Y/N name \$/ML \$ \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40 20	usage 3.80 ainfall data ar expected wate use MI/ha 2.7 4.2	pumping cost 11.16 ad therefore al water use from model MI/ha 1.70 1.63	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/Mil 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2	stre based of ice tonne \$145.00 \$320.00	on these av V Costs \$/ha \$338.85 \$387.78	verage data. Gross Margir \$/ha \$386.15 \$252.22	Totəl GM \$15,4 \$5,0
n-farm recycling system cheme details name usage charge fixed charge fare resource cost regulated supplies unregulated supplies groundwater supplies ote : Water Use from model given be b. frrigated enterprises finter crops Short Fallow - Wheat Long Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins Ummer crops Cotton Sorghum Rice Soybeans Maize Sunflowers	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40 20	usage 3.80 sinfall data ar expected wate use MI/ha 2.7 4.2	pumping cost 11.16 ad therefore al water use from model MVha 1.70 1.63	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pri tonnes/ha \$/h 5 2	stre based o ice tonne \$145.00 \$320.00	on these av ∨ Costs \$/ha \$338.85 \$387.78	verage data. Gross Margir \$/ha \$386.15 \$252.22	Total GM \$15,4 \$5,0
In-farm recycling system cheme details name usage charge fixed charge fixed charge valer resource cost regulated supplies groundwater supplies ote : Water Use from model given be B. Irrigated enterprises Finter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Soybeans Maize Soundowers	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40 20	usage 3.80 sinfall data ar expected wate use MVha 2.7 4.2	pumping cost 11.16 ad therefore al water use from model Mi/ha 1.70 1.63	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/ 5 2	re based (ice ionne \$145.00 \$320.00	on these av ∨ Costs \$/ha \$338.85 \$387.78	verage data. Gross Margir \$/ha \$386.15 \$252.22	Total GM \$15,4 \$5,0
n-farm recycling system cheme details name usage charge fixed charge fixed charge fixed charge after resource cost regulated supplies groundwater supplies ote : Water Use from model given be p. firfigated enterprises Finter crops Shof Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Soybeans Maize Sunflowers Soture Luceme bax	Y/N name \$/ML \$ \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40 20	usage 3.80 ainfall data an expected wate use MI/ha 2.7 4.2 4.2	pumping cost 11.16 nd therefore al water use from model MI/ha 1.70 1.63	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/Mil 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2	stan 57	on these av V Costs \$/ha \$338.85 \$387.78 \$387.78	verage data. Gross Margir \$/ha \$386.15 \$252.22	Total GM \$15,4 \$5,0
In-farm recycling system cheme details name usage charge fixed charge Vater resource cost regulated supplies unregulated supplies groundwater supplies ote : Water Use from model given be 3. Irrigated enterprises finter crops finter crops finter dilow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Soybeans Maize Sunflowers asture Luceme hay Perennial nasture (Luceme)	Y/N name \$/ML \$ \$/ML \$/ML elow is for 97 year aver	14.96 age monthly ra Area Ha 40 20 indication of sto 80	usage 3.80 ainfall data ar expected wate use MI/ha 2.7 4.2 4.2 6.2	pumping cost 11.16 ad therefore al water use from model MI/ha 1.70 1.63	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2 963	sre based i ice tonne \$145.00 \$320.00 \$320.00	on these av V Costs \$/ha \$338.85 \$387.78 \$387.78 \$387.08	verage data. Gross Margir \$/ha \$386.15 \$252.22 \$252.22	Total GM \$15,4 \$5,0 \$35,1
n-farm recycling system cheme details name usage charge fixed charge fixed charge diater resource cost regulated supplies groundwater supplies ote : Water Use from model given be b. Irrigated enterprises Tinter crops Shot Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Sorghum Rice Sunflowers sature Luceme hay Perennial pasture (Lucerne) - String	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver [U.SM's/ha or an	14.96 age monthly ra Area Ha 40 20 20	usage 3.80 ainfall data ar expected wate use MI/ha 2.7 4.2 4.2 cking rate) 8.5	pumping cost 11.16 nd therefore al water use from model MVha 1.70 1.63 4.85	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2 9.63	state.57	on these av ∨ Costs \$/ha \$338.85 \$387.78 \$387.78	verage data. Gross Margir \$/ha \$386.15 \$252.22 \$252.22	Total GM \$15,4 \$5,0 \$35,1
In-farm recycling system cheme details name usage charge fixed charge valer resource cost regulated supplies groundwater supplies ote : Water Use from model given be 2. Irrigated enterprises Finter crops Short Fallow - Wheat Long Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Soybeans Maize Sunflowers asture Luceme hay Perennial pasture (Luceme) - Spring	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver (LSM's/ha or an	14.96 age monthly ra Area Ha 40 20 indication of sto 80	usage 3.80 ainfall data ar expected wate use MI/ha 2.7 4.2 4.2 bcking rate) 8.5	pumping cost 11.16 ad therefore al water use from model MI/ha 1.70 1.63 4.85	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$// 5 2 9.63	state	on these av ∨ Costs \$/ha \$338.85 \$387.78 \$387.78 \$991.08	rerage data. Gross Margir \$/ha \$386.15 \$252.22 \$438.92	Total GM \$15,4 \$5,0 \$35,1
h-farm recycling system Scheme details name usage charge fixed charge Valer resource cost regulated supplies groundwater supplies ote : Water Use from model given be B. Irrigated enterprises Vinter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Soybeans Maize Sunflowers asture Lucerne hay Prevennial pasture (Lucerne) - Spring - Summer - Autumn	Y/N name \$/ML \$ \$/ML \$/ML elow is for 97 year aver [U.SM's/ha or an	14.96 age monthly ra Area Ha 40 20 indication of sto 80	usage 3.80 ninfall data ar expected wate use MI/ha 2.7 4.2 4.2 bcking rate) 8.5	pumping cost 11.16 ad therefore al water use from model MI/ha 1.70 1.63	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/Mil 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2 9.63	state state of the	on these av V Costs \$/ha \$338.85 \$387.78 \$387.78 \$3991.08	verage data. Gross Margir \$/ha \$386.15 \$252.22 \$438.92	Total GM \$15,4 \$5,0 \$35,1
In-farm recycling system cheme details name usage charge fixed charge Yater resource cost regulated supplies groundwater supplies ote : Water Use from model given be B. Irrigated enterprises Tinter crops Short Fallow - Wheat Long Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Soybeans Maize Sunflowers asture Luceme hay Perennial pasture (Luceme) - Spring - Summer - Autumn	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver (LSM's/ha or an	14.96 age monthly ra Area Ha 40 20 indication of sto 80	usage 3.80 ninfall data ar expected wate use MI/ha 2.7 4.2 4.2 bcking rate) 8.5	pumping cost 11.16 ad therefore al water use from model MVha 1.70 1.63 4.85	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2 9.63	state state of the	on these av ∨ Costs \$/ha \$338.85 \$387.78 \$387.78 \$991.08	yerage data. Gross Margir \$/ha \$386.15 \$252.22 \$252.22	Total GN \$15,4 \$5,0 \$35,1
In-farm recycling system cheme details name usage charge fixed charge valer resource cost regulated supplies groundwater supplies ote : Water Use from model given be B. Irrigated enterprises Finter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Soybeans Maize Soundowers asture Lucerne hay Perennial pasture (Lucerne) - Spring - Summer - Autumn - Winter Annual pasture (sub clover)	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver (LSM's/ha or ar	14.96 age monthly ra Area Ha 40 20 indication of sto 80	usage 3.80 sinfall data ar expected wate use MVha 2.7 4.2 4.2 bocking rate) 8.5	pumping cost 11.16 ad therefore al water use from model MVha 1.70 1.63 4.85 2.12	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2 9.63 *	state	on these av ∨ Costs \$/ha \$338.85 \$387.78 \$387.78 \$991.08 \$991.08	verage data. Gross Margir \$/ha \$386.15 \$252.22 \$438.92 \$438.92	Total GN \$15,4 \$5,0 \$35,1 \$35,1
n-farm recycling system cheme details name usage charge fixed charge fixed charge fixed charge fixed charge fixed supplies groundwater supplies ote : Water Use from model given be c. Irrigated enterprises Finter crops Shof Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Soybeans Maize Sunflowers asture Lucerne hay Perennial pasture (Lucerne) - Spring - Summer - Auturnn - Winter Annual pasture (sub clover) - Spring	Y/N name \$/ML \$ \$/ML \$/ML elow is for 97 year aver [U.SM's/ha or an	14.96 age monthly ra Area Ha 40 20 indication of sto 80	usage 3.80 ainfall data ar expected wate use Ml/ha 2.7 4.2 4.2 bcking rate) 8.5	pumping cost 11.16 nd therefore al water use from model MI/ha 1.70 1.63 4.85	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/Mil 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2 9.63 Iink Cattle Budge	state state in the state of the	on these av V Costs \$/ha \$338.85 \$387.78 \$991.08 \$991.08	verage data. Gross Margir \$/ha \$386.15 \$252.22 \$438.92 \$438.92	Totəl GM \$15,4 \$5,0 \$35,1 \$35,1
n-farm recycling system cherme details name usage charge fixed charge fixed charge fixed charge fixed supplies unregulated supplies groundwater supplies ote : Water Use from model given be . Irrigated enterprises finter crops finter crops Stont Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Soybeans Maize Sunflowers sature Luceme hay Perennial pasture (Luceme) - Spring - Summer - Auturnn - Winter Annual pasture (sub clover) - Spring - Summer	Y/N name \$/ML \$ \$/ML \$/ML elow is for 97 year aver (LSM's/ha or ar	14.96 age monthly ra Area Ha 40 20 indication of sto 80	usage 3.80 ainfall data ar expected wate use MI/ha 2.7 4.2 4.2 ocking rate) 8.5	pumping cost 11.16 ad therefore al water use from model MI/ha 1.70 1.63 4.85 2.12	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2 9.63 [*] Iink Cattle Budge	stre based i ice tonne \$145.00 \$320.00 \$148.57	on these av V Costs \$/ha \$338.85 \$387.78 \$387.78 \$3991.08 \$334.85	verage data. Gross Margir \$/ha \$386.15 \$252.22 \$438.92 \$438.92	Total GM \$15,4 \$5,0 \$35,1 \$12,5
n-farm recycling system cherme details name usage charge fixed charge fixed charge fixed charge fixed charge after resource cost regulated supplies ote : Water Use from model given be c. Irrigated enterprises Tinter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Soybeans Maize Sounfowers Sunfowers Sunfowers Sunfowers Sunfowers Sunfowers Sunfowers Sunfowers Sunfowers Sunfowers Sunfowers Sunfowers Sunfowers Sunfowers Sunfowers Sunfowers Sunfowers Summer - Autumn Winter Autumn	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver (LSM's/ha or ar (LSM's/ha or ar \$/DSE	14.96 age monthly ra Area Ha 40 20 indication of sto 80	usage 3.80 sinfall data ar expected wate use MVha 2.7 4.2 4.2 ocking rate) 8.5	pumping cost 11.16 ad therefore al water use from model MVha 1.70 1.63 4.85 2.12	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2 9.63 Iink Cattle Budge	state	on these av ∨ Costs \$/ha \$338.85 \$387.78 \$991.08 \$991.08	yerage data. Gross Margir \$/ha \$386.15 \$252.22 \$438.92 \$438.92 \$627.89	Total GN \$15,4 \$5,0 \$35,1 \$12,5
In-farm recycling system cheme details name usage charge fixed charge valer resource cost regulated supplies groundwater supplies ote : Water Use from model given be chart fallow - Wheat Long Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Soybeans Maize Sunflowers ssture Luceme hay Perennial pasture (Luceme) - Spring - Summer - Autumn - Winter Sourier Summer - Autumn	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver (LSM's/ha or an (LSM's/ha or an Gross Margin \$/DSE	14.96 age monthly ra Area Ha 40 20 indication of sto 80	usage 3.80 ainfall data ar expected wate use MI/ha 2.7 4.2 4.2 bcking rate) 8.5	pumping cost 11.16 ad therefore all water use from model MI/ha 1.70 1.63 4.85 2.12	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/Mil 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2 9.63 Iink Cattle Budge	state	on these av ∨ Costs \$/ha \$338.85 \$387.78 \$991.08 \$334.85	verage data. Gross Margir \$/ha \$386.15 \$252.22 \$438.92 \$438.92 \$627.88	Total GN \$15,4 \$5,0 \$35,1 \$12,50
h-farm recycling system Scheme details name usage charge fixed charge Valer resource cost regulated supplies groundwater supplies ote : Water Use from model given be 2. Irrigated enterprises Finter crops Short Fallow - Wheat Long Fallow - Wheat Long Fallow - Wheat Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Soybeans Maize Sunflowers Sunflowers Sunflowers Sunflowers Sunflowers - Spring - Summer - Autumn - Winter Autumn - Winter - Autumn - Autumn - Autumn - Autumn - Autumn - Autumn - Autumn - Autumn - Autumn - Autumn	Y/N name \$/ML \$ \$/ML \$/ML elow is for 97 year aver (LSM's/ha or an (LSM's/ha or an Gross Margin \$/DSE	14.96 age monthly ra Area Ha 40 20 indication of str 80	usage 3.80 ainfall data ar expected wate use MI/ha 2.7 4.2 4.2 bocking rate) 8.5	pumping cost 11.16 ad therefore al water use from model MI/ha 1.70 1.63 4.85 2.12	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/Mil 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2 9.63 [*] Iink Cattle Budge	state state in the second state of the second	on these av V Costs \$/ha \$338.85 \$387.78 \$991.08 \$334.85	yerage data. Gross Margir \$/ha \$386.15 \$252.22 \$438.92 \$438.92 \$627.88	Total GM \$15,4 \$5,0 \$35,1 \$12,5 \$12,5
In-farm recycling system cheme details name usage charge fixed charge dived charge diver resource cost regulated supplies groundwater supplies ote: Water Use from model given be P. Irrigated enterprises Finter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Sorghum Rice Sunflowers sature Lucerne hay Perennial pasture (Lucerne) - Spring - Summer - Auturnn - Winter Summer - Auturnn - Winter Summer - Auturnn - Winter Summer - Auturnn - Winter Other crops	Y/N name \$/ML \$ \$/ML \$/ML s/ML elow is for 97 year aver (LSM's/ha or ar (LSM's/ha or ar Gross Margin \$/DSE	14.96 age monthly ra Area Ha 40 20 indication of sto 80	usage 3.80 ainfall data ar expected wate use MI/ha 2.7 4.2 4.2 bocking rate) 8.5	pumping cost 11.16 ad therefore al water use from model MI/ha 1.70 1.63 4.85 2.12	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2 9.63 ³ Iink Cattle Budge	state	on these av V Costs \$/ha \$338.85 \$387.78 \$387.78 \$991.08 \$334.85	verage data. Gross Margir \$/ha \$386.15 \$252.22 \$438.92 \$438.92	Total GM \$15,4 \$5,0 \$35,1 \$12,50 \$12,50
h-farm recycling system Scheme details name usage charge fixed charge Valer resource cost regulated supplies groundwater supplies ote : Water Use from model given be R. Irrigated enterprises Finter crops Short Fallow - Wheat Long Fallow - Wheat Barley Oats Canola Chickpeas Fababeans Lupins ummer crops Cotton Sorghum Rice Soybeans Maize Sunflowers asture Luceme hay Perennial pasture (Luceme) - Spring - Summer - Autumn - Winter Annual pasture (sub clover) - Spring - Summer - Autumn - Winter Miner Other crops tal Water Use (ML)	Y/N name \$/ML \$ \$/ML \$/ML \$/ML elow is for 97 year aver (LSM's/ha or ar (LSM's/ha or ar Gross Margin \$/DSE	14.96 age monthly ra Area Ha 40 20 20 indication of sto 80 20	usage 3.80 ainfall data ar expected wate use MVha 2.7 4.2 4.2 bcking rate) 8.5 3 3	pumping cost 11.16 ad therefore al water use from model MI/ha 1.70 1.63 4.85 2.12 2.12	fixed cost 3.07 I Gross Marg Pump/Deliver Cost \$/MI 14.96 14.96	in figures here a Yield Pr tonnes/ha \$/1 5 2 9.63 Iink Cattle Budge	state	on these av ∨ Costs \$/ha \$338.85 \$387.78 \$991.08 \$334.85	yerage data. Gross Margir \$/ha \$386.15 \$252.22 \$438.92 \$438.92 \$627.88	Total GM \$15,4 \$5,0 \$5,0 \$35,110\$\$35,110\$\$\$35,110\$\$\$35,110\$\$\$35,110\$\$\$35,110\$\$\$35,110\$\$\$35,110\$\$\$35,110\$\$\$\$35,110\$\$\$\$35,110\$\$\$\$\$35,110\$

1.2 – Zone 2 Far	m Continued
------------------	-------------

Vinter crops							Area	Yield		V.Costs	Gross Margir	Total G
Short Fallow - Wheat								100	22	#1C0.02	£150.07	175
Long Fallow - Wheat								160	2.2	\$160.03	\$150.97	\$20
Date .												
- Canola								160	1.5	\$350.14	\$129.86	\$20
- Chickpeas												
- Fababeans												
- Lupins												
Cotton												
Sorahum												
Soybeans												
Maize												
Sunflowers												
'asture	(LSM's	s/ha or an	indication of st	tocking rate)				20	4.00	\$166 DA	\$214.00	
Luceme hay -establishment				\$14.43				20 280 link St	4.00	\$160.50	\$147.81	\$41
- Spring				Gross Margin								
- Summer				\$/DSE								
- Autumn												
- Winter												
Annual pasture (sub clover)												
- Spring												
- Summer												
- Autumn												
Other crops (Fallow)								20				
Pryland Gross Margin								640 Ha			_	\$93
The second												
5. Overhead cost structure												
and constructions of the set of t												
Administration expenses												
accounting		2000										
bank charges		1800										
insurance (farm & Vehicles)		3000	150		ACCETC						E	
super & workers compensation		5497	15%		ASSETS	the transmission was		A	je j	urrent Value	Expected Life	
telephone		800	of wages		Form					1750000		
stationary		UCC ANAAF	Station Hand (343						1750000		
uel and oil (farm vehicles)		5200	Station manu C	50 J	Tractor - 150 k	n				00003		
lectricity (not including pumping costs)		1200			Tractor - 100 h	p				45000		
Penairs and maintenance		1200			Tractor - 75 hp	P						
plant and equipment		5000			Tractor - MF 3	5						
structures		2000			Tractor with Lo	ader - 50 hp				25000		
Depreciation			RATE:		Header					50000		
plant and equipment		28390	6.8%		Mover / Condit	ioner -Two				46000		
- structures					Rake - Two					10000		
lates					Rake - Two					6000		
Stocking charges (PPB)		300								15000		
Land		2500			Baler					15000		
Water		3070			Sprayer	14/				15000		
other operating overheads (Rego & License)		1200			Bale Loader of	wagon				20000		
otal operating overneads	•	99,155			Field Bins - Tu	0				8000		
					Plough	iu .				10000		
Drefit and Financial analysi					Flough					50000		
5. Profit and Financial analysis	S				Scarifier			-		5000		
and the second					Combine					15000		
THER FARM INCOME (eg timber)	\$				Harrows - Two					12000		
		100 000			Motor Bike					5000		
OTAL FARM GROSS MARGIN	5	162,062			Iruck					17600		
OTAL OPEDATING OVERUEADS		00 152			Ute - Two	tor Plant				17500		
QTAL OPERATING OVERHEADS	>	99,153			Travelling Irriga	tor Plant			-			
	5	62 909										
ess operators labour	S	24,000			Sundries					30000		
									-	417500		
PERATING RETURN (A)	5	38,909			Livestock	Number	Value/	hd				
ess			P component		Rams -	45	\$100)	4500			
OD interest paid	\$	2,800			Ewes -	1500	\$30		45000			
HP / Lease interest paid	5	6,650	\$11,581		Bulls -	1	\$1,00	0	1000			
Term Loan interest	5	21,250	\$8,855		Cows -	50	\$450		22500			
		0 100			Liquid accete	uch as						
USINESS RETURN (B)		0,209			Bank Denosite	. Off farm inv	ectment	24	5 000 -	unor		
OTAL ASSETS (C)					Shares / Equit	v	ounion		20000	apor		
liquid assets	5	45.000										
value of land and improvements	\$ 1.	750,000										
value of plant and equipment	5	417,500			LIABILITIES					Rate	Term	
value of sheep	\$	49,500			OD Bank				80000	10.5%	1	
value of cattle	\$	23,500			HP / Lease				70000	9.5%	5	
ub Total	\$ 2.	285,500			Mortgage			2	50000	8.5%	15	
						2 1 .			0.0			
OTAL LIABILITIES	\$	400,000				Bank lending	equity		81%			
QUITY (D)	\$ 1,1	885,500										
		00 501										
QUITY RATIO D/C X 100		82.5%										
FTUDN ON TOTAL ACCETC (A) (/O 400		4 70										
ETURN ON FOULTY (P) / (D) (- 400		1.7%										
		0.4%										
	and go a	25.000										
		25,000										
F-FARM INCOME	2	32 200										
F-FARM INCOME prox. Taxable Farm Profit or Loss	5	32,209										
FF.FARM INCOME prox. Taxable Farm Profit or Loss ET CASH RESULT (after tax)	\$ \$ \$	32,209 58,721										
FF-FARM INCOME pprox. Taxable Farm Profit or Loss ET CASH RESULT (after tax) FICION TREES :	5 5 5	32,209 58,721										

1.3 – Zone 3 Small Farm

1. Physical farm characteristi	ics	as r onio	an 100 /0 01	, uniter of	Zone 5 - 30 fa	armers				
arm areas										
otal farm area	Ha	1000	range 400-500	0						
wea set up for irrigation	Ha	200	range 40-2000							
vea normally irrigated	Ha	200	one watering o	in some pasture						
ryland area	Ha	800		l						
eef cattle number	no.	40	cows							
heep number	no.	1200	First X ewes							
ator cumpline						Farm "nlan" sh	owing rotation	(clackwise)		
allet supplies					50% F & 50%	Graded	ioning rotation	(clocitinoc)		
imention entitlement	MI	072	range 400.800	n	50 /011 0 50 /0	Irrigated 200 H	Orviand 800 ha		1000	
ingation entitiement	NAL	512	range 400-000	0		O 36 ha unda	Eollow 100 ha		ha	
access to on allocation	IAIL				75.0	14/ (C 2)	1 allow - 100 Ha			
iregulated water					10.0	W (C /)	001		Nac	
icenced area	Ha				425.0	W	PP-Legume		W	
oundwater supplies						W (undersown	7)	1		
rrigation entitlement	ML				225.0	L establishme	PP-Legume		Canola	B under
osest rainfall site	location	Condobolin	rain 15-16 inch	ies or 8-30 cm		L (3 yrs)				
					150.0	AP-S (4 yrs)	PP-Legume		PP-Legume	
rm labour						AP-S (4 yrs)				
owner/manager	no. of weeks	50			875.0	Total water use	9			
amily	no. of weeks									
ernament labour	no, of weeks	48								
anian about	no of weeks									
			2 neonle				Merino - Jamhin	n March / Ar	nil X D	1
			r hookie				22.23 Micron 2	25 Micron	5 Kg vield 65%	
(Lo micron 2		- g Jield 00 /0	
irrigation characteristics							Age	# Ewes		
		Area	Irrigation eff				2.5	1	2% mortality	
igation by layout and method		Ha	%				3.5	1	1.5 % joining	
and formed - flood (LFF)		100	80				4 5	i i	100 % selling	Contraction of the local distance
and formed - spray (LES)			00				5.5	1		
and landformed - flood (NI FE)		100	60				Total	1200	_	
an landformed - nood (NEFT)		100	00				Total	Sheer July	Aug	
ion ianulumeu - spray (NLFS)						L		Sheet July	Aug	
JITEI										
igation infra-structure										
ver pump details										
capacity	MI/hour	0.5	12ML/day							
running cost	\$/hour	6	\$12 / ML							
ound water pump details										
capacity	MI/hour									
nunning cost	\$/hour									
form storage	, vinour									
auface cros	cause matrac									
denth	square metres									
depth	metres	N	Mark and dear	L						
i-farm recycling system	Y/N	N	high cost of rec	cycling						
cheme details		1								
name	name	JID								
usage charge	\$/ML									
fixed charge	\$	10 to 20								
ater resource cost			usage	pumping cost	fixed cost					
regulated supplies	\$/ML	13.10	3.80	9.30	3.07					
unregulated supplies	\$/ML					•				
proundwater supplies	\$/ML									
2										
te · Water lise from model riven helow	e for 97 year aver	ane monthly ra	infall data an	d therefore all	Gross Margin	figures here a	re based on th	ese averane	data	
te : water use from model given below i	s iui si year aver	age monuny is	illiali uata ali	u mereiore an	GIUSS Maryin	inguies nere a	ie based on di	ese average	uala.	
Irrigated enterprises		Area	expected water	water use from	Pump/Delivery	Yield I	Price	V.Costs	Gross Margin	Total C
		11-			C 604	A	14	5 Aug	F/L-	
		па	use minna	порег мила	COST #/MI	tonnes/ha	MOIIIIG	¢/∏d	ψ/lid	
nter crops										
Short Fallow - Wheat										
ong Fallow - Wheat		75	3.00	2.92	13.10	4	\$145.00	\$434.35	\$145.65	\$10
larley										
lats	undersowing co	25	3.00	2.92	13.10	2.50	\$120.00	\$172.94	\$127.06	5
anola	to cattle	and the second								
hickneas	25 DSE/ha for 3	months								
ahahaans	Autumn Jamhing	1								
avavčalij		3								
սիութ										
mmer crops										
otton										
sorghum										
lice										
loybeans										
faize										
unflowers										
ture	(LSM's/ha or an	indication of sto	ocking rate)							
ucerne hav		EU.	8.50	6 4E	13 10	11.00	\$128.00	\$1 003 12	\$404 88	\$20
erennial nacture (Lucerne)	3 DSE nor Ac	50	0.50	0.40	13.10	,1.00	+120.00	+1,000.12	+ 104.00	720
Sering	7 5 DOE DE AC									
· opning	7.5 USE/ha									
Summer										
Autumn										
Winter										
nnual pasture (sub clover)	4 DSE per Ac	50	3.00	2.55	13.10	link Cattle Bude	gets	\$198.01	\$106.37	\$5
Spring	Spring Jamhing	~	0.00	2.00				1.00.01		
Summer	fat lamba cold									
	Sections 5010	10								
Autumn	Sept/Oct 18-20	<g< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></g<>								
Winter										
ther crops										
	1									
I Water Use (ML)		200 Ha	875.0 ML	742.6 ML						
al Water Use (ML)		200 Ha	875.0 ML	742.6 ML						
al Water Use (ML) gated Gross Margin		200 Ha	875.0 ML	742.6 ML						\$39

1.3 – Zone 3 Small Farm Continued

4. Dryland enterprises										
and a second						A.02	Viold	VCaets	Groce Margin	Total GM
Wheat after pacture (Pixer Block)						Area 100		\$231.11	\$58.89	101al Gr
- Wheat into stubble (River Block)						100	2	\$138.49	\$141.51	\$14.1
Barley						100		\$150.45	•1-1.01	
· Oats										
- Canola			undersowing	cost to sheep		100	1.2	\$301.32	\$82.68	\$8,2
- Chickpeas										
Fababeans										
- Lupins	1									
and the second										
Summer crops										
Cotton										
- Sorghum										
Soybeans										
Maize										
Sunflowers										
asture	(LSM's/ha or	an indicat	ion of stocking rate)							
Luceme hay								470 50		
Perennial pasture (Lucerne)	4 DSE/Ha					400) link Sheep2X Bu	\$73.56	\$109.74	\$43
- Spring	1600 ewes ed	uvalent a	ll up							
- Summer	12 - 16 DSE 1	or 4U cow	S							
- Autumn										
- Winter										
Annual pasture (sub clover)										
- Spring										
- Summer										
- Autumn										
- Winter						100				
Uther crops (Fallow)						100	,			
										\$77
Oryland Gross Margin						800) Ha			\$12.
5. Overhead cost structure										
dministration expenses									350 + 250 Lavou	ıt
accounting	140	n							400 dry	
hank charges	180	n							250 impronveme	nt
insurance (farm & Vehicles)	210	ñ							Loo in provining	
euner & workers compensation	330	7	15%	ASSETS			Age	Current Value	Expected Life	
telephone	80	n a	anew j	ASSETS	-		- Age	ouncil tuide	Expected Ene	
ctationary	51	ິ	Huges	Farm				1250000		
shour (normanant & cacual)	226	IG Station	Hand Gd 3	PLANT				1 .200000		
abour (permanent & casual)	2204			Tractor - 150 h	hn - ID		10	25000		
le and oli (rarm venicles)	100	N N		Tin truck Roy	lp -JU		35	2000		
ectricity (not including pumping costs)	120	N		Placher	uiuiu		10	5000		
repairs and maintenance	201	n		Siastier			40	3000		
- plant and equipment	200	0		Pergie	front loador		40	10000		
structures	120	U DA	TE	OU NP MF 65 0	C front loader		30	20000		
Depreciation		RA	IE:	Header - NH1:	550		25	2000		
- plant and equipment	1424	6 6.8	3%	Group			10	5000		
- structures				Workshop Eq	uipment		~~~~	1000		
Rates		-		Pump 8" 50 h	p - Perkins		20	5000		
- Stocking charges (PPB)	31			MAN FI	011			12000		
- Land	123	0 \$ 1.30	/ AC	venicie - Faic	on CILI		4	5000		
- Water	290	14		Boom Spraye	3r		15	5000		
Other operating overheads (Rego & License)	80	U		Harrows	a second and		10	5000		
otal operating overheads	\$ 60,25	3		Augers	2		30	5000		
				Field Bins	3 @ 18 tons		20	3000		
				Plough			-			
6. Profit and Financial analysis				Scarifier - 211	Inter		15	5000		
				Combine - 28r	nm Sheaner		15	10000		
THED EADM INCOME (on timber)	s .			Motor Bike	4WD		2	6500		
THER FARM INCOME (BY UNDER)				Motor Bike	2W . Yamaha	200	2	1000		
OTAL FADM CDOSS MADCIN	£ 111.0C	, January and States		Truck Inter	299 - Tallialia	200	25	40000		
UTAL FARM GRUSS MARGIN	a 111,00	3		Truck - Inter			7	15000		
				Ote - Hilux 2.8			1	7000		
OTAL OPERATING OVERHEADS	\$ 60,25	3		UTF-sets 28 pla	ate - 3.5 Inter		15	/000		
				OTOLIOTUDE						
IET FARM INCOME	\$ 51,61			STRUCTURES	S			10000		
ess operators labour	\$ 24,00	J to JUU	U	Sundries				10000		
	l son same							209500		
PERATING RETURN (A)	\$ 27,61	5		Livestock	Number	Value/hd				
ess	1	P com	oonent	Rams -	20	\$100	2000			
OD interest paid	\$ 3,50]		Ewes -	1200	\$30	36000			
HP / Lease interest paid	\$ 3,80]	\$6,617	Bulls -	1	\$1,000	1000			
Term Loan interest	\$ 6,800)	\$2,834	Cows -	40	\$450	18000			
	la companya				1					
USINESS RETURN (B)	\$ 13,51	5		Liquid assets	such as					
				Bank Deposits	s - Off farm inves	tment	25,000	super		
UTAL ASSETS (C)	Lan Lot granges			Shares / Equit	Y	ļ				
liquid assets	\$ 25,000					ļ				
value of land and improvements	\$ 1,250,000	1								
value of plant and equipment	\$ 209,500)		LIABILITIES				Rate	Term	
value of sheep	\$ 38,000)		OD Bank			100000	10.5%	1	
	\$ 19,000)		HP / Lease			40000	9.5%	5	
value of cattle	\$ 1,541,500)		Mortgage	1		80000	8.5%	15	
value of cattle ub Total					ļ					
value of cattle ub Total										
value of cattle ub Total OTAL LIABILITIES	\$ 220,000]								
value of cattle ub Total OTAL LIABILITIES	\$ 220,000	J								
value of cattle ub Total OTAL LIABILITIES QUITY (D)	\$ 220,000)								
value of cattle ub Total OTAL LIABILITIES QUITY (D)	\$ 220,000 \$ 1,321,500)								
value of cattle ub Total OTAL LIABILITIES QUITY (D) QUITY RATIO D/C X 100	\$ 220,000 \$ 1,321,500 85.74)) %	80							
value of cattle ub Total OTAL LIABILITIES QUITY (D) QUITY RATIO D/C X 100	\$ 220,000 \$ 1,321,500 85.71)) %	80							
value of cattle ub Total OTAL LIABILITIES QUITY (D) QUITY RATIO D/C X 100 ETHRN ON TOTAL ASSETS (A) / (C) = 100	\$ 220,00 \$ 1,321,500 85.79) %	80							
value of cattle ub Total OTAL LIABILITIES QUITY (D) QUITY RATIO D/C X 100 ETURN ON TOTAL ASSETS (A) / (C) x 100 ETURN ON TOTAL ASSETS (A) / (C) x 100	\$ 220,000 \$ 1,321,500 85.79 1.89) , , , , , , , , , , , , , , , , , , ,	80							
value of cattle ub Total OTAL LIABILITIES QUITY (D) QUITY RATIO D/C X 100 ETURN ON TOTAL ASSETS (A) / (C) x 100 ETURN ON EQUITY (B) / (D) x 100	\$ 220,000 \$ 1,321,500 85.79 1.89 1.09)) 6 	80							
value of cattle ub Total OTAL LIABILITIES QUITY (D) QUITY RATIO D/C X 100 ETURN ON TOTAL ASSETS (A) / (C) × 100 ETURN ON EQUITY (B) / (D) × 100	\$ 220,000 \$ 1,321,500 85.75 1.85 1.05	5 6	80							
value of cattle ub Total OTAL LIABILITIES QUITY (D) QUITY RATIO D/C X 100 ETURN ON TOTAL ASSETS (A) / (C) x 100 ETURN ON TOTAL ASSETS (A) / (C) x 100 FF-FARM INCOME	\$ 220,000 \$ 1,321,500 85.71 1.81 1.01 \$ 18,000	6 6	80							
value of cattle b Total DTAL LIABILITIES QUITY (D) QUITY RATIO D/C X 100 ETURN ON TOTAL ASSETS (A) / (C) x 100 ETURN ON EQUITY (B) / (D) x 100 EFF-FARM INCOME IPTOX. Taxable Farm Profit or Loss TO ADU DEGUIT (Concurs)	\$ 220,00 \$ 1,321,500 85,71 1.81 1.05 \$ 18,000 \$ 37,515	6 6 6	80							
value of cattle ub Total DTAL LIABILITIES QUITY (D) QUITY RATIO D/C X 100 ETURN ON TOTAL ASSETS (A) / (C) x 100 ETURN ON EQUITY (B) / (D) x 100 FF-FARM INCOME IPTOX. Taxable Farm Profit or Loss ET CASH RESULT (after tax)	\$ 220,00 \$ 1,321,50 85.79 1.89 1.09 \$ 18,000 \$ 37,519 \$ 52,807	5 6 6 6	80							
value of cattle ub Total OTAL LIABILITIES QUITY (D) QUITY RATIO D/C X 100 ETURN ON TOTAL ASSETS (A) / (C) x 100 ETURN ON EQUITY (B) / (D) x 100 FF-FARM INCOME oprox. Taxable Farm Profit or Loss ET CASH RESULT (after tax) ECICION TREES -	\$ 220,00 \$ 1,321,50 85,79 1,89 1,99 \$ 18,000 \$ 18,000 \$ 18,000 \$ 37,519 \$ 52,807		80							

1.4 – Zone 3 Large Farm

1. Physical farm characterist Farm areas Total farm area Area set up for irrigation Area normally irrigated Dryland area Beef catle number Sheep number Water supplies Regulated water - irrigation entitlement - access to off allocation Unregulated water - licenced area	Ha Ha Ha Ha Ha no. no.	5000			7000 E 20 fo					
Farm areas Total farm area Area set up for irrigation Area normally irrigated Dryland area Beef cattle number Sheep number Water supplies Regulated water - access to off allocation Unregulated water - licenced area	Ha Ha Ha Ha no. no.	5000			Z016.2 - 30 1a	rmers				
Total farm area Area set up for irrigation Area normally irrigated Dryland area Beef cattle number Sheep number Water supplies Regulated water - arccess to off allocation Unregulated water - licenced area	Ha Ha Ha Ha no. no.	5000		~						
Area set up tor ringation Area normally irrigated Dryland area Beef cattle number Sheep number Water supplies Regulated water - inrigation entitlement - access to off allocation Unregulated water - licenced area	на На На no. no.	2000	range 2000-200	W						
Vice rounding ingred Dryland area Beef cattle number Sheep number Water supplies Regulated water - irrigation entitlement - access to off allocation Unregulated water - licenced area	Ha no. no.	200	one watering or	some pasture						
Union area Beef cattle number Sheep number Water supplies Regulated water - irrigation entitlement - access to off allocation Unregulated water - licenced area	no. no.	4800	one watering of	i some pastore						
Water supplies Regulated water - irrigation entitlement - access to off allocation Unregulated water - licenced area	no.	200	cows - range 10	0-1200, mostly	on dryland (finis	hed on irrigat	ion)			
Water supplies Regulated water - irrigation entitlement - access to off allocation Unregulated water - licenced area		2000	M ewes on dryl	and						
Water supplies Regulated water - irrigation entitlement - access to off allocation Unregulated water - licenced area	an a	1200	1X ewes on irric	ated		Farm "plan"	showing rotat	ion (clockwise)	
Regulated water - irrigation entitlement - access to off allocation Unregulated water - licenced area					50%LF & 50%	Graded				
- irrigation entitlement - access to off allocation Unregulated water - licenced area						Irrigated 200	Dryland 800	ha		
- access to off allocation Unregulated water - licenced area	ML	972	range 400-8000		60.0	W-20 ha	Fallow -100 I	ha	W	
Unregulated water - licenced area	ML				60.0	O (undersow	n)	<u></u>		
 licenced area 					90.0	AP-S	PP-Legume		**	
	Ha				390.0	AP-S			Canala	Rundaran
Groundwater supplies	L.					AP-S	PP-Legume		Califola	D dildersom
- imgation entitlement	ML	Condoholin	rain 15 16 inch	e or 8.30 cm		AP-S	PP-I entime		PP-I equine	
Closest rainfall site	location	CONGODORI	Tain 15-10 men	5 01 0-50 cm		AP-S	L' Legomo		Logenie	
Carm Jahour						AP-S	-			
- owner/manager	no of weeks	50				AP-S	a second cardenes			
- family	no. of weeks	50			600.0	Total water u	ISP			
- pernament labour	no. of weeks	48								
- casual	no. of weeks	48			4000 ha	F- 400 ha	W	W	C - 200 ha	
			4 people		Dryland				undersown	
									O -200 ha	Home
2. Irrigation characteristics									undersown	Timber
		Area	Irrigation eff			PP - Legume	PP - Legume	PP - Legume	PP - Legume	Native veg
rrigation by layout and method		Ha	%							Hill
- land formed - flood (LFF)		100	80				1			
- land formed - spray (LFS)										
- non landformed - flood (NLFF)		100	60			200 Cows &	2000 Ewes			
- non landformed - spray (NLFS)										
- other										1
							Merino - lan	nbing March /	April X D	
rrigation infra-structure							22-23 Micror	22.5 Micron	т- 6 Kg yield 65	0%
River pump details			1014111				Age	# Ewes	2% montality	
- capacity	MI/hour	0.5	12ML/day				2.5		2 % monality	
- running cost	\$/hour	ь	\$12/ML				3.5		1.5 % jurning	(Determine)
Fround water pump details	Milhour						4.5		too // sening	
- capacity	Mi/hour						Total	1200		
- running cost	annour						, oral	Sheer July /	Aug	
- surface area	square metres									
- depth	metres									
On-farm recycling system	Y/N	N	high cost of rec	ycling						
Scheme details										
- name	name	JID								
- usage charge	\$/ML									
- fixed charge	\$									
Water resource cost			usage	pumping cost	fixed cost					
- regulated supplies	\$/ML	13.10	3.80	9.30	3.07					
- unregulated supplies	\$/ML									
- groundwater supplies	⊅/IVIL									
Note · Water lise from model river below	is for 97 year aver	age monthly r	ainfall data an	d therefore all	Gross Margin f	igures here	are based or	n these avera	ge data.	
note . Hater ose nom moder graen below		- Je monuny i								TALO
3. Irrigated enterprises		Area	expected water	water use from	Pump/Delivery	Yield	Price	V.Costs	Gross Margin	Total GM
		Ha	use Ml/ha	model Ml/ha	Cost \$/MI	tonnes/ha	\$/tonne	\$/ha	\$/ha	
Winter crops										
- Short Fallow - Wheat			and the	and the second	1. 2.5					
- Long Fallow - Wheat		20	3.00	2.92	13.10	4	\$145.00	\$434.35	\$145.65	\$2,9
- Barley			8.4 HL	And the second						
- Oats		20	3.00	2.92	13.10	2.50	\$120.00	\$172.94	\$127.06	\$2,5
- Canola										hanne ann an
Chickpeas										
Chickpeas Fababeans										
Chickpeas Fababeans Lupins										
Chickpeas Fababeans Lupins										
Unickpeas Fababeans Lupins Summer crops										
- Chickpeas Fababeans - Lupins Summer crops - Cotton - Sondhum										
Uhrckpeas Fababeans Lupins Cotton Sorghum Rice										
Chickpeas Fababeans Lupins Summer crops Cotton Sorghum Rice Sovbeans										
Chickpeas Fababeans Lupins Summer crops Cotton Sorghum Rice Soybeans Maize										
Chickpeas Fababeans Lupins iummer crops Cotton Sorghum Rice Soybeans Maize Sunflowers										
Chickpeas Fababeans Lupins commer crops Cotton Sorghum Rice Soybeans Maize Sunflowers										
Chickpeas Fababeans Lupins iummer crops Cotton Sorghum Rice Soybeans Marze Sunflowers asture	(LSM's/ha or ar	indication of st	tocking rate)							
Chickpeas Fababeans Fababeans Lupins Summer crops Cotton Sorghum Rice Soybeans Maize Sunflowers Pasture Luceme hay	(LSM's/ha or ar	indication of st	tocking rate)							
Chickpeas Fababeans Fababeans Lupins summer crops Cotton Sorghum Rice Soybeans Maize Sunflowers Pasture Lucerne hay Perennial pasture	(LSM's/ha or ar	indication of st	tocking rate)							
Chickpeas Fababeans Lupins Summer crops Cotton Cotton Sorghum Rice Soybeans Maize Sunflowers Pasture Luceme hay Perennial pasture Spring	(LSM's/ha or ar	indication of st	tocking rate)							
Chickpeas Fababeans Fababeans Lupins Summer crops Cotton Sorghum Rice Soybeans Maize Sunflowers Pasture Lucerne hay Perennial pasture Sing Sumer	(LSM's/ha or ar	indication of st	tocking rate)							
Chickpeas Fababeans Fababeans Lupins Summer crops Cotton Sorghum Rice Soybeans Maize Sunflowers Pasture Lucerne hay Perennial pasture Sying Summer Autumn	(LSM's/ha or ar) indication of st	locking rate)							
Chickpeas Fababeans Lupins conton Sorghum Rice Soybeans Maize Sunflowers Casture Lucerne hay Perennial pasture - Spring - Summer - Autumn - Muter	(LSM's/ha or ar) indication of st	tocking rate)							
Chickpeas Fababeans Lupins iummer crops Cotton Sorghum Rice Soybeans Marze Sunflowers Suture Luceme hay Perennial pasture - Spring - Summer - Autumn - Autumn - Winter Annual pasture (sub clover)	(LSM's/ha or ar	indication of st	tocking rate) 3.00	2.55	13.10	link Cattle B	udgets	\$495.94	\$2,033.14	\$60,9
Chickpeas Fababeans Lupins Summer crops Cotton Sorghum Rice Soybeans Maize Sunflowers Asture Lucerne hay Perennial pasture Summer - Spring - Summer - Autumn - Winter Annual pasture (sub clover) Annual pasture (sub clover)	(LSM's/ha or ar	indication of st 30 130	tocking rate) 3.00 3.00	2.55 2.55	13.10 13.10	link Cattle B link Sheep22	udgets { Budgets	\$495.94 \$229.23	\$2,033.14 \$258.81	\$60,9 \$33,6
Chickpeas Fababeans Lupins Summer crops Cotton Sorghum Rice Soybeans Maize Sunflowers Pasture Luceme hay Perennial pasture Spring Summer Antual pasture (sub clover) Annual pasture (sub clover) Annual pasture (sub clover) Summer	(LSM's/ha or ar	indication of st 30 130	tocking rate) 3.00 3.00	2.55 2.55	13.10 13.10	link Cattle B link Sheep2)	udgets < Budgets	\$495.94 \$229.23	\$2,033.14 \$258.81	\$00 \$33,6
Chickpeas Fababeans Lupins Summer crops Cotton Sorghum Rice Soybeans Maize Sunflowers Pasture Luceme hay Perennial pasture - Spring - Summer - Autumn Winter Annual pasture (sub clover) Annual pasture (sub clover) - Summer - Summer - Autumn - Autumn - Summer - Autum - Summer - Summer - Autum - Summer -	(LSM's/ha or ar	indication of st	tocking rate) 3.00 3.00	2.55 2.55	13.10 13.10	link Cattle B link Sheep22	udgets K Budgets	\$495.94 \$229.23	\$2,033.14 \$258.81	\$60,9 \$33,6
Chickpeas Fababeans Fababeans Lupins Summer crops Cotton Sorghum Rice Soybeans Maize Sunflowers Pasture Lucerne hay Perenial pasture Summer Autumn Annual pasture (sub clover) Annual pasture (sub clover) Summer - Autumn - Mutter Autumn - Winter	(LSM's/ha or ar	indication of st 30 130	tocking rate) 3.00 3.00	2.55 2.55	13.10 13.10	link Cattle B link Sheep2)	udgets (Budgets	\$495.94 \$229.23	\$2,033.14 \$258.81	\$60,9 \$33,6
Chickpeas Fababeans Lupins Summer crops Cotton Sorghum Rice Soybeans Maize Sunflowers Auturn Winter Auturn Winter Other crops	(LSM's/ha or ar	indication of st 30 130	locking rate) 3.00 3.00	2.55 2.55	13.10 13.10	link Cattle B link Sheep2)	udgets (Budgets	\$495.94 \$229.23	\$2,033.14 \$258.81	\$60,9% \$33,6
Chickpeas Fababeans Lupins container Sorghum Rice Soybeans Maize Sunflowers stature Lucerne hay Perennial pasture - Spring - Summer - Autumn - Winter Annual pasture (sub clover) - Annual pasture (sub clover) - Summer - Autumn - Winter - Summer - Autumn - Winter - Summer - Autumn - Winter - Summer	(LSM's/ha or ar	indication of st 30 130	a.oo 3.00 3.00	2.55 2.55	13.10 13.10	link Cattle B link Sheep2	udgets (Budgets	\$495.94 \$229.23	\$2,033.14 \$258.81	\$60,9 \$33,6
Chickpeas Fababeans Lupins iummer crops Cotton Sorghum Rice Soybeans Maize Sunflowers 'asture Luceme hay Perennial pasture - Spring - Summer - Autumn - Winter Annual pasture (sub clover) Annual pasture (sub clover) - Summer - Summer - Autumn - Summer - Summer - Summer - Summer - Summer - Summer - Summer - Summer - Autumn - Winter Other crops tal Water Use (ML)	(LSM's/ha or ar	indication of st 30 130 200 Ha	tocking rate) 3.00 3.00 600.0 ML	2.55 2.55 524.5 ML	13.10 13.10	link Cattle B link Sheep2	udgets Budgets	\$495.94 \$229.23	\$2,033.14 \$258.81	\$60,9 \$33,6

4. Dryland enterprises	I Determination		a nast [] is no it max till no metropyce((4))and		ung H Millertri II. I. Soorthers	100 TO 1 1000		ger freit Alle Art Kolle-A	
						No.14	¥0	0	Tatalona
Winter crops Wheat after pacture (Piver Block)					Area	Yield	V.Costs	Gross Margin \$58.89	1 otal GM \$5,889
- Wheat into stubble (River Block)					100		2 \$138.49	\$141.51	\$14,151
- Wheat after pasture (Dry Block)					400	1.5	5 \$199.76	\$10.24	\$4,094
- Wheat into stubble (Dry Block)			underson inc. excet to show M		400	1.5	5 \$130.84	\$86.66	\$34,662
- Canola (River Block)			undersowing cost to sheep?		100	1.3	2 \$301.32	\$82.68	\$8,268
- Canola (Dry Block)			undersowing cost to sheepM		200	1.0	\$238.69	\$81.31	\$16,263
- Lupins									\$0 60
Summer crops									\$0 \$0
- Cotton									\$0
- Sorghum									\$0
- Soybeans - Maize									90 \$0
- Sunflowers									\$0
Pasturo	(I SM's/ba or ar	indication of s	tocking rate)						\$0 \$0
- Luceme hay	(LONIS/IIa OF al	indication of s	locking jare)						\$0
- Perennial pasture (Legume Mix)					2000	link SheepN	\$19.66	\$10.14	\$20,271
- Spring									\$U \$0
- Autumn									\$0
- Winter									\$0
- Annual pasture (sub clover)									\$U \$0
- Summer									02 02
- Autumn									\$0
- Winter - Other crops (Fallow)					500				07 12
- Home, Timber, Native Veg, Hill					800				~
Dryland Gross Margin					4800	Ha			\$131,033
5 Overhead cost structure									
o, oremean cost structure								350 + 250 Layou	ıt 1
Administration expenses	-							400 dry	
- accounting	2500							250 improoveme	nt
- insurance (farm & Vehicles)	2100							a mi + 250K mi	prouvernerits
- super & workers compensation	7429	15%	ASSETS			Age	Current Value	Expected Life	
- telephone	900	of wages	Farm				2500000		
Labour (permanent & casual)	49526	Station Hand (Gd 3 PLANT :				250000		
Fuel and oil (farm vehicles)	2800		Tractor - 300 h	p - 4WD Case		2	150000		
Electricity (not including pumping costs) Renairs and maintenance	1200		Lip truck - Bed Slasher	ford		35	5000		
- plant and equipment	2800		Fergie			40	3000		
- structures	1200		80 hp MF 65 C	front loader		30	10000		
Depreciation	50490	68%	Group	ase 2180		10	5000		
- structures			Workshop Equ	ipment			10000		
Rates	200		Pump 8" 50 hp	- Perkins		20	5000		
- Stocking charges (PPB)	13250	\$ 1.00 / Ac	Vehicle - Land	cruiser		4	41000		
- Water	2984	and the second	Compu Spray	er		15	18000		
Other operating overheads (Rego & License)	\$ 140 629		Harrows	2		10	5000		
Total operating overneads	4 140,020		Field Bins	5 @ 30 tons		20	10000		
			Chisel Plough				20000		
6. Profit and Financial analysis			Scarifier - 211	Inter		15	5000		
OTHER FARM INCOME (eg timber)	s .		Motor Bike	4WD		15	6500		
officient (eg uniber)	·		Motor Bike	2W - Yamaha	200	2	1000		
TOTAL FARM GROSS MARGIN	\$ 231,127		Truck - Inter	20		15	50000		
TOTAL OPERATING OVERHEADS	\$ 140 629		Off-sets 60 pla	20 te - 3.5 Inter		15	30000		
			AirSeeder - Fle	xicod			100000		
NET FARM INCOME	\$ 90,498		STRUCTURES	-			10000		
	✓ 40,000		Sundnes				742500		
OPERATING RETURN (A)	\$ 42,498	_	Livestock	Number	Value/hd				
Less - OD interest paid	\$ 7.000		Rams - Ewes -	75	\$100	7500 96000			
- HP / Lease interest paid	\$ 19,000	\$33,087	Bulls -	4	\$1,000	4000			
- Term Loan interest	\$ 8,500	\$3,542	Cows -	200	\$450	90000			
BUSINESS RETURN (B)	\$ 7,998		Liquid assets a	uch as					
			Bank Deposits	- Off farm investi	ment	50,000	super		
TOTAL ASSETS (C)	¢ 60.000		Shares / Equit	1					
- value of land and improvements	\$ 2,500,000								
- value of plant and equipment	\$ 742,500		LIABILITIES				Rate	Term	
- value of sheep	\$ 103,500		OD Bank			200000	10.5%	1 F	
Sub Total	\$ 3,490,000		Mortgage			100000	9.5%	5 15	
	1								
TOTAL LIABILITIES	\$ 500,000								
EQUITY (D)	\$ 2,990,000								
								1	
EQUITY RATIO D/C X 100	85.7%	85							
RETURN ON TOTAL ASSETS (A) / (C) × 100	1.2%								
RETURN ON EQUITY (B) / (D) × 100	0.3%								
OFF-FARM INCOME	\$ 18.000								
Approx. Taxable Farm Profit or Loss	\$ 55,998								
NET CASH RESULT (after tax)	\$ 76,659								
DECISION TREES :-				1					

1.4 – Zone 3 Large Farm Continued

1.5 – Zone 4 Farm

Cotton / Maize + Summer Crops 1. Physical farm characteristics	- 30 Farm 5	iers								
Farm areas										
Total farm area	Ha	7500	range 2000-	50000						
Area set up for irrigation	Ha	1300 8	600 ha Floo	id, 500 ha Spray,	200 ha to deve	lop				
Area normally irrigated	Ha	550	450 ha Floo	id, 100 ha Spray			For Int F	havi i		
Dryland area	Ha	6950		4 6			Farm "plan" s	showing rotat	lion	
Beef cattle number	no.	2075	50 COWS &	2 75 tomo			1200 ho cot i		6200 ha dayland	
Sneep number	10	1000	Self renlacir	& 75 failts			flood irrinated	. 450 ha	0200 na urgiano	
Water sunnlies		1000	merino to 1)	X (from m flock)		180 0	- Long Fallow	w - Wheat		
Regulated water		1000 1		(ion in lock)		1500.6	- Cotton - Co	onventional	1.	
- irrigation entitlement	ML	4000	with 2000 M	1L conjunctive with	h GW	374.9	- Cotton - GN	M -spray irriga	ated	
- access to off allocation	ML					1800.0	- Maize - Flo	od		
Unregulated water - licenced area	На					360.0	- Perennial p	asture (Lucer	me)	1.000
Groundwater supplies										
- conjunctive use entitlement	ML	2000							and the second	a service
- base entitlement	ML	2000							A CONTRACTOR	
Total Water availablity	ML	6000					150 ha		3000 sheep : 20	00 Merino Ev
Closest rainfall site	location	Hillston							(1000M & 1000) BL)
Farm labour							spray irrigate	d - 100 ha	2.5 % rams	
- owner/manager	no. of weeks	50							50 cows - 350	kg@\$1.10
- family	no. of weeks	50							- 85% replace	ment
- pernament labour	no. of weeks	48							U1 bull	
- casual	no. of weeks								1.000	
O funimenties a strenged in st							100		1.00	
2. Irrigation characteristics							400 ha			
		Area I	rrigation eff							
Irrigation by layout and method		Ha	%							
- land formed - flood (LFF)			8	RO			Tedat	0.64		
- land tormed - spray (LFS)							10 develop 20	iu ha		100
- non landformed - flood (NLFF)				b5			L			
- non landformed - spray (NLFS)				90		ADIE E MI	Total water w			
- other						4215.5 IVIL	. Tutai water us	Se		
Irrigation infra structure								Sheen -> 2	3 Micron Wool -	6 Kg
River numn details								Lambing >	> 80% M	ong
- capacity	Ml/hour	1 875 4	15ML/day -	20" pump and a b	ackup 12" pun	np as well		100% - Grif	fith Sale Yard -	\$ 60
- running cost	\$/hour	14.06	\$7.50	per ML	and the first	P		2% mortality	1	
Ground water pump details								Maize stubbl	e -> finished or	Lucerne
- capacity	Ml/hour	0.625 1	15ML/day							
- running cost	\$/hour	\$13.75	\$22.00	per ML				Oats - 20 h	a every 2 years	
On-farm storage								water if nece	essary	
- surface area	square metres									
- depth	metres			in the substitute						
On-farm recycling system	Y/N	Y 2	25% re-cycl	e possibility ?						
Scheme details										
- usage charge	\$/ML									
- fixed charge	\$			numping cost	fixed cost					
water resource cost	E/NAL	11.24	3 BU		3.07					
- regulated supplies	\$/IVIL	11.24	3.00	7.44	3.07					
- unregulated supplies	\$/IVIL \$/MI	22.30	0.30	22.00	0.61					
- groundwater supplies	\$/MI	21.00	0.00	21.00	0.01					
- spray inigation costs (extra)	WINE_	21.00		21.00						
Note : Water Use from model given below is f	or 97 year aver	rage monthly rai	infall data	and therefore a	II Gross Margi	n figures her	e are based o	on these ave	rage data.	
3. Irrigated enterprises		Area e	expected wa	atewater use from	Pump/Delivery	Yield	Price	V.Costs	Gross Margin	Total GM
		Ha	use MI/ha	model Ml/ha	Cost \$/MI	tonnes/ha	\$/tonne	\$/ha	\$/ha	
Winter crops										
- Short Fallow - Wheat										\$0
- Long Fallow - Wheat		60.0	3	.0 3.1	11.24	3.5	145.00	\$424.75	\$82.75	\$4,965
- Barley										\$0
- Oats										\$0
- Canola										\$0
- Unickpeas										\$U 60
- Fababeans										90 00
- Lupins										0. 0.
Summer crone										م
- Cotton - Conventional		187 5	8	0 113	11.74	6.60 Lint	450.00	\$2,287.57	\$945.75	\$177.328
		107.5	0		11.24	2.38 Seed	110.00	10,000,01	10.10	\$0
- Cotton - GM -spray irrigated		62.5	6	.0 10.0	32.24	6.60 Lint	450.00	\$2,309.43	\$923.88	\$57,743
					and the second sec	2.38 Seed	110.00			\$0
- Maize - Flood		180.0	10.	.0 8.0	11.24	10.00	160.00	\$859.91	\$740.09	\$133,216
- Maize - Spray										\$0
										\$0
Pasture	(LSM's/ha or an	n indication of sto	cking rate)							\$0
- Lucerne hay						ALL STREET		To the second		\$0
- Perennial pasture (Lucerne)		60.0	An an		No. of Concession, Name	11.00	On farm Use -	cost transfer	rred to livestock	\$0
Self replacing Merino 21-22 Micron		15	6.	U 9.6	11.24	link SheepM	Budgets	\$1,154.54	\$855.95	\$12,839
First X Lambs		33	6.	U 9.6	11.24	link Sheep1X	Budgets	\$664.28	\$1,0/6.37	\$35,520
Cattle		12	6.	U 9.6	11.24	iink Cattle Bu	lugets	\$966.50	\$685.31	\$8,224
Annual posturo (cub stand)										\$0
- Annual pasture (sub clover)										0¢
- Shunda										Diệ Diệ
- Summuc										9U 60
- Autumn Winter										an Uite
- Other crops										0. 0
										~
Total Water Use (ML)		550 Ha	4215.5 M	L 4936.6 ML		6				
										00.) 0100 000000000
Irrigated Gross Margin										\$429,835
									-	

Appendices

1.5 – Zone 4 Farm Continued

4. Dryland enterprises					1 				
Winter crops					Area	Yield	V.Costs	Gross Margin	Total GM
- Short Fallow - Wheat							-		\$0
- Long Fallow - Wheat After Pasture					150) 1.5	\$215.86	\$1.64	\$246
- Long Fallow - Wheat After Stubble		undersowing	g cost to cattle		150	1.5	\$127.84	\$89.66	\$13,448
- Uats									U¢ N2
- Canola - Chickness									\$0
- Fababeans									\$0
- Lupins									\$0
									\$0
Summer crops									\$0
- Cotton									\$0
- Sovheans									\$0
- Maize									\$0
- Sunflowers									\$0
	0.000	industrial of sheeting sets)							\$0
Pasture	(LOM SITIA OF A	i indication of stocking rate)							\$0
- Perennial (native) pasture		maintennan	ce costs to overh	eads-Weed con	5900	for cattle & s	heep		\$0
- Spring									\$0
- Summer									\$U \$0
- Autumn									30
- Annual pasture (sub clover)									\$0
- Spring									\$0
- Summer									\$0
- Autumn									\$0 #0
- vvinter					750	1			50
other clops (r allow + to develop)					7.50				~ 0
Dryland Gross Margin					6950	Ha			\$13,694
				600 ha laid out	flood 200 las	ered @ 800/A	c = 400,000		
						400	@ 400/Ac =	400,000	
5. Overhead cost structure				100 ha spray				= 350,000	
				3000 ML river				= 100,000	
Administration expenses				3000 MI GW				- 620.000	
- accounting	5000			b200 dryland			=	- 820,000 230.000	
- bank charges	1000			Another	and all a set the set			= 900.000	
- super & workers compensation	3724	15%	ASSETS	L		Age	Current Value	Expected Life	
- telephone	3000	of wages							
- stationary	3000		Farm				3000000		
Labour (permanent & casual)	24825	Rural Tradesperson	Tractor - Case	MX 270		1	145000		
Electricity (not including pumping costs)	5000		Tractor - Case	7110 / bucket		4	70000		
Repairs and maintenance			Fordson 5000				7000		
- plant and equipment	5000	50% GM ?	Fiat 1000				7000		
- structures	11000	8 - 14.5 K	Header NHIR	87		1	20000		
Dryland maintenance	48797	68%	Augers - 9" 33	15		4	2800		
Landforming - 50 ha @ \$750	37500		Augers - 7" 44			12	1800		
Rates			Chaser Bin - 8	lt		4	8000		
- Stocking charges (PPB)	300		Total Inter A	1020		15	15000		
- Land Water (both surface and GW)	13500	rates TUK PPD TK	Grouper	1030		15	7500		
Other operating overheads - (Rego & License)	2500		Air Seeder - C	ommor Shea		12	15000		
Total operating overheads	\$ 204,946		Chiesel plough	n - Shearer 25'		5	20000		
			4WD Motor Bi	ke		2	1000		
0. Bus fit and Financial analysis			MOTOR DIKE			2	1000		
6. Prom and Financial analysis			Mulcher - Meh	way		8	10000		
OTHER FARM INCOME (ag timber)	s .		Ripper - Agro J	21		4	2000		
(agistment from good seasons included in off farm	income)		Lister / fertilise	er - Twin Rivers		2	25000		
TOTAL FARM GROSS MARGIN	\$ 443,529		Tanks				8000		
			Cultivator - too	lbar - 8m		5	8000		
TOTAL OPERATING OVERHEADS	\$ 204,946		Boom Spray -	Mardi UD		10	3500		
NET FARM INCOME	\$ 238 584		Bed Shaper			2	5000		
Less operators labour	\$ 48,000		Cultivator - Rol	lling 8 row		1	25000	2	
			Rotabuck			5	2000		
OPERATING RETURN (A)	\$ 190,584	D comparent	Syphons (2800	J)			10000		
Less	\$ 9.750	includes OD + Cron terms	Ute - Cruiser	· ND		2	30000		
- HP / Lease interest paid	\$ 24,000	\$51,137	Hilux			1	25000		
- Term Loan interest	\$ 29,100	\$9,669	Hilux			7	6000		
			Commodore			2	20000		
BUSINESS RETURN (B)	\$ 165,234		STRUCTURES	o Arico			25000		
TOTAL ASSETS (C)			workshop Sur	iules			717606		
- liquid assets	\$ 200.000		Livestock	Number	Value/hd				
- value of land and improvements	\$ 3,000,000		Rams -	75	\$100	7500			
- value of plant and equipment	\$ 717,606		Ewes -	3000	\$30	90000			
- value of sattle	 97,500 37,600 		Dulls -	50	\$1,000	22500			
Sub Total	\$ 4 038 606		COM2 -	50	\$4.00	22300			
			Liquid assets :	such as					
TOTAL LIABILITIES	\$ 750,000		Bank Deposits	- Off farm hous	6	150,000	plus super		
FOUNDY (D)	r 2 200 000		Shares / Equit	У		50000			
	\$ 3,268,606								
EQUITY RATIO D/C X 100	81.4%		LIABILITIES				Rate	Term	
			OD Bank			150000	11.50%	1	
	1 70/		HP / Lease			300000	8%	5	
RETURN ON TOTAL ASSETS (A) / (C) × 100	4.1%					300000	9.70%	15	
RETURN ON TOTAL ASSETS (A) / (C) × 100 RETURN ON EQUITY (B) / (D) × 100	4.7%		Mortgage			200000	0.04		
RETURN ON TOTAL ASSETS (A) / (C) × 100 RETURN ON EQUITY (B) / (D) × 100 DEE EARM INCOME (including contenent)	4.7% 5.0%		Mortgage Crop Terms			200000	8% 10.5%		
RETURN ON TOTAL ASSETS (A) / (C) × 100 RETURN ON EQUITY (B) / (D) × 100 OFF-FARM INCOME (including agistment) Approx. Taxable Farm Profit or Loss	4.7% 5.0% \$ 25,000 \$ 213.234		Mortgage Crop Terms			200000	8% 10.5% 9.5%		
RETURN ON TOTAL ASSETS (A) / (C) × 100 RETURN ON EQUITY (B) / (D) × 100 OFF-FARM INCOME (including agistment) Approx. Taxable Farm Profit or Loss NET CASH RESULT (after tax)	4.7% 5.0% \$ 25,000 \$ 213,234 \$ 146,078		Mortgage Crop Terms			200000	8% 10.5% 9.5% 8.5%		

1.6 – Zone 5 Farm

Summer Crops - 25 Farmers					Zone 5 - 15 fa	armers				
1 Dhucical farm characterist	ics				7000 3 - 10 f	armore				
I. Filysical faim characterist	103				20110 5 - 101	anners				
Farm areas	На	2000	range 600-5000							
Area set up for impation	Ha	750	range 320-1900	/ developing	on average 10 h	na/pa				
Area normally irrigated	Ha	350	plus Pasture if	water available	e	and the second second second second				
Dryland area	На	1750								
Beef cattle number	no.	150	150 Angus Cow	rs & 4 Bulls						
Sheep number	no.	1700	link to LSM ? N	LEwes 900 h	a (including non	rarable 300)				
Water supplies					Farm "plan" sl	howing rotation				
Regulated water		4.000	000 000		H. Sain	dentes for initiation	1			loop arable
- irrigation entitlement	ML	1400	range 800-2600		normally irriga	t setup for irrigation		devland 1200	lha	non arable
- access to off allocation	ML	U	U%	150.0	350 ha	400 ha	0 . 100 ha	hav	w .	50 na
Unregulated water	На			210.0	010	W	0 - 100 ha		"	
- incenced area Groupdwater supplies	1 ld			200.0	AP (seed hav	C	С	В	C	
- impation entitlement	ML				M	PP-Luc/Medic			101000	
Closest rainfall site	location	Jemalong Wei	r	800.0	M	PP-Luc/Medic	AP	AP	F	
		or ?:	Condobolin		PP-Luc/Medic	AP-Sub		1.	Spray AP	
Farm labour				270.0	PP-Luc/Medic	AP-Sub	AP	AP	AP	
- owner/manager	no. of weeks	50				devt				
- family	no. of weeks									
- pernament labour	no. of weeks	48		1630.0) Total water us	EDryland Rotation:				
- casual	no. of weeks	10				C(200ha)-W(100ha)- B(100ha) and V	V(200ha) und	ersown to AP(3 y	15)
						AP = lucerne with anr	nual pasture	Elve F	O He esseister e	41.
						oats not undersown -	more likely wheat	- more like 3	o na specialty p	UK
2. Irrigation characteristics						- grazed & stripped				
		Area	Irrigation eff							
Irrigation by layout and method		Ha	%							
- land formed - flood (LFF)		500	80							
- land formed - spray (LFS)		050	05							
- non landformed - flood (NLFF)		250	65							
- non landformed - spray (NLFS)										
- otner										
Irrigation infra structure										
River numn details										
- canacity	Ml/hour									
- running cost	\$/hour									
Ground water pump details										
- capacity	Ml/hour									
- running cost	\$/hour									
On-farm storage										
- surface area	square metres									
- depth	metres									
On-farm recycling system	Y/N									
Scheme details		IID								
- name	name ¢/M	JID								
- usage charge	a) IVIL									
- lixeo charge Water resource cost			Govt usage	III useage	Govt Fixded	JIL Fixed				
- regulated supplies	\$/ML	10.57	3.80	6.77	3.07	6.50				
- unregulated supplies	\$/ML	10101	4.62	5.95	3.09	6.50	from JIL Feb 200	1		
- groundwater supplies	\$/ML									
3										
Note : Water Use from model given below	is for 97 year avera	age monthly r	ainfall data and	l therefore a	II Gross Margi	n figures here are ba	sed on these av	erage data.		
2 Irrigated enterprises		A	avpacted wate	unter une from	- Dumn/Dolivory	Viold	Drice	VCoete	Gross Marnin	Total GM
5. migaled enterprises		Area	expected watev	valer use iron	Pump/Delivery	Tielu	Frice	V.CU515	Closs Margin	Total Offi
		Ha	use Ml/ha r	nodel Ml/ha	Cost \$/MI	tonnes/ha	\$/tonne	\$/ha	\$/ha	
Winter crops										-
- Short Fallow - Wheat					40.57		140.00	E100 4	enco co	\$0
- Long Fallow - Wheat		50	3.0	2.9	10.57	5.0	140.00	\$4.39.40	\$260.60	\$13,030
- Darley										50 50
- Oals		50	12	2.9	10.57	2.4	300.00	\$394 40	\$325.51	\$16 275
- Chickneas		30	4.2	2.0	10.57	2.4	3.0.00	400-1.10	4020.01	\$0
- Fababeans										\$0
- Lupins										\$0
International Contraction of the										\$0
Summer crops										\$0
- Cotton										\$0
- Sorghum										\$0
- Rice										\$0
- Soybeans				_			100.00			\$0
- Maize		100	8.0	7.8	10.57	10.0	160.00	\$895.62	\$/04.38	\$/0,438
- Sunflowers										\$U 60
Destars	0 514-0	indiaction of t	ocking ret-							ېن ۵
Pasture	(LSM's/ha or an	indication of st	ocking rate)							* 0
- Lucerne nay	DD J (M	100	27	1 9	10.57	link Cattle & chaan h	idnets - 50% each	\$596 17	\$99.18	\$9 918
- Lucerne / Wedic Masture	2 / t/ba bay	100	2.1	1.8	10.57	min Carne or sneep DL	agera - Ju % edCl	4030,17	\$33.10	02,010 02
- Summer	2.4 Una nay 3 dse / 6 month	s								\$0
- Summer	3 926 1 Q 110UU									\$0
- Minter										50
- Annual pasture (sub clover)	AP - HW	50	40	25	10.57	link Cattle Budgets		\$473.81	\$103.99	\$5,200
- Spring	HW-500 kg/ha 6	a \$3.20	4.5	2.0	10.01					\$0
- Summer	L- 500-750 ka/h	а								\$0
- Autumn	Feb-Aug 5dse/h	a								\$0
- Winter		1 A & B								\$0
- Other crops										\$0
Total Water Use (ML)		350 Ha	1630.0 ML	1372.6 ML						
										6114 004
Irrigated Gross Margin						1	1			a 114,001

1.6 – Zone 5 Farm *Continued*

Winter crops				A	lea	Yield	V.Costs	Gross Margin	Total (
- Short Fallow - Wheat									
- Long Fallow - Wheat			undersowing of	cost to sheep	30	00 2	2.0 \$231.11	\$58.89	\$17
- Barley					11	00 1	1.8 \$145.13	\$70.87	\$/
- Oats					11	50 1	1.0 Un farm Use	- cost \$205/ha tr	ansierrec
- Canola Chickness					2.	50	•250.05	401.51	420
- Fababeans									
- Lupins									
Summer crops									
- Cotton									
- Sorghum									
- Soybeans									
Maize									
Sumowers									
Pasture	(LSM's/ha or an	indication of stocking rat	te)						
- Lucerne hay -establishment									
- Perennial pasture									
- Spring									
- Summer									
- Autumn									
- winter	4 dse / ha	Only esta	ablishment costs - u	undersowing + fe	60	00 link Sheep Bu	udaets		
ub pasture (on land laid out for irrigation)		no mainte	enance cost	3	10	00 link Sheep Bu	udgets		
egume pasture (lucerne + medic)					10	00 link Sheep Bu	idgets		
- Autumn									
- Winter									
- Other crops - Fallow					10	00			
· being developed / non-arable					1(50 Ha		-	E 41
ryland Gross Margin					1/5	зо па			34
b. Overnead cost structure								[roo	0.000
dministration expenses								500 ha lasered (2 \$2500
accounting	1400							1500 ha imgable (@ \$100
bank charges	1800							1500 na urylanu	a sau
super & workers compensation	2100	15%	ASSETS			Ane	Current Value	Expected Life	
super & workers compensation	4004	of wares	ASSETS	and the second descent of the		Age	Current Value	L'Apecieu Lie	
stationary	550		Farm				2650000		
abour (permanent & casual)	30425	Rural Tradesperson	PLANT :						
uel and oil (farm vehicles)	2800		J.D. Tractor or	n HP - 300s		2	110000		
lectricity (not including pumping costs)	1200	•	Header on Lea	ase - NHTR 87		3	150000		
lepairs and maintenance			Air seeder on	HP		2	80000		
plant and equipment	2800		Gestner Tool I	Bar & Tools			20000		
structures	4000	24500	Planter - JD, H	linze			10000		
Depreciation	20700	RATE:	Gas gear on L	lister ng			16000		
plant and equipment	12250	0.070	Griedv off eet	e 64 disc			22000		
andforming - 25na (2) \$490	12230		Harrows 12 m	5 04 0150			2000		
Stocking charges (PPB)	300								
Land	1230	4.16 / ha	Slasher				1000		
Water	13398		Ford 5000 c F	ront loader			12000		
other operating overheads - (Rego & License)	800		Case 120 Hp			5	50000		
otal operating overheads	\$ 120,197		Chaser Bin			-	2000		
			Field Bins x 2	45' x 301			8000		
Profit and Einanaial analysis			Augers X 2 (40 x 30)		2	5000		
b. PIOIN allo Fillancial allalysis			4 VV DIKE			2	15000		
TUED FADM INCOME (an dimbed			4 WD Ute			5	9000		
THER FARM INCOME (eg timber)	•		4 WD Ole			3	12000		
OTAL FARM GROSS MARGIN	\$ 159 943		Truck - Inter -	inreg			5000		
	• 100,010		Grouper				12000		
OTAL OPERATING OVERHEADS	\$ 120,197								
ET FARM INCOME	\$ 39,747		STRUCTURES	5			47000		
ess operators labour	\$ 24,000		Sundries				15000		
PERATING RETURN (A)	\$ 15.747		Livestock	Number	Value/hd		505000		
ess	- 13,141	P component	Rams -	50	\$100	500	00		
OD interest paid	\$ 5,250		Ewes -	1700	\$30	5100	00		
HP / Lease interest paid	\$ 11,400	\$19,852	Bulls -	4	\$1,000	400	00		
Term Loan interest	\$ 15,300	\$6,376	Cows -	150	\$450	6750	00		
USINESS RETURN (P)	5 (3053)		stasse hiuni l	such as					
USINE SS RETURN (D)	• (3,553)		Bank Deposits	s - Off farm investn	nent	25,00	0 super		
OTAL ASSETS (C)			Shares / Equit	ty / house		120,00	U		
liquid assets	\$ 145,000								
value of land and improvements	\$ 2,650,000						Data	Tarm	
value of plant and equipment	3 585,000					150.00	10 5%	1	
value of cattle	\$ 71.500		HP / Lease			120.00	0 9.5%	5	
ub Total	\$ 3,507,500		Mortgage / Tel	rm Loan		180,00	0 8.5%	15	
OTAL LIABILITIES	\$ 450.000								
	\$ 3,057,600								
(0) THUS	• 3,037,500			\$ 50 K Drawings					
	87.2%	85		permanent labour casual 11.80 / hr	\$28-30 K / pa				
QUITY RATIO D/C X 100				earthworks maint	enance \$ 10 K				
ETURN ON TOTAL ASSETS (A) / (C) x 100	0.4%			Contribution in anti-					
ETURN ON TOTAL ASSETS (A) / (C) × 100 ETURN ON EQUITY (B) / (D) × 100	0.4% -0.1%			lasering 50 ha @	490 / ha				
QUITY RATIO D/C X 100 ETURN ON TOTAL ASSETS (A) / (C) × 100 ETURN ON EQUITY (B) / (D) × 100	0.4% -0.1%			lasering 50 ha @	490 / ha				
2011Y RATIO D/C X 100 ETURN ON TOTAL ASSETS (A) / (C) x 100 ETURN ON EQUITY (B) / (D) x 100 F.F.ARM INCOME	0.4% -0.1% \$ 15,000			lasering 50 ha Q	490 / ha				
2017Y RATIO D/C X 100 ETURN ON TOTAL ASSETS (A) / (C) x 100 ETURN ON EQUITY (B) / (D) x 100 FFFARM INCOME prox. Taxable Farm Profit or Loss	0.4% -0.1% \$ 15,000 \$ 20,047			lasering 50 ha @	490 / ha				

Appendix 2: Data used in the hydrology calculations

Hydrolog	iy - Lac	chlan Val	lley																							1																	
Zone 4-1	Hillston						rrigation V	Vater Recu	irement Ru	Cron and	ivestock 0	(/ha)	_				-		-																								
	Raintall	Riverflow F	armstorage	Groundwate Total Wate	Asperadus	Torratoes	Sweet	Oats	Wheat	Critico	Catton	Weite	Maize	Lucerne	Incerne	Derennial	Sit.				20. 20																						
Month	mm			Available			com			conventional	GM - Spray	Find	Soray	hav	nasture	ractive	ciouer	Livestock	1 westork	21 westnow																							
July	31.3	3			0	0	0	0	0.024	0	1	1	0	101	poside	howne	COIG	UVESIUCA	TUVESIULA	LUVESIDEN	1																						
August	30.5	5			0	0	0	0129	0.420	0	0	0	0	0	A	0																											
September	26.3	3			0	0	0	0.535	1 007	n	i i		0	0	0 382	ň																											<u>.</u>
October	35.9	9			0	0	0	0.638	1 186	0 161	E143	0	0	0.626	1 398	0.245		1																									
November	26.5	5			0	0.35	0 322	0.030	0.490	1.130	1004	0.855	0.760	0.020	1.330	0.343		1																									
December	30.5	5			0.755	1.62	1 392	0.001	0.100	2 472	2107	2124	1 907	1 167	1,100	0.23/																											
January	30.0	0			0.735	2.01	1 007	0	0	2.972	2.13/	2.010	1.03/	1.107	1.003	0.705																											
Fohnier	20.2	•			0.000	1.03	1.307	0	0	3.007	2.0/3	2313	6.000	1.102	1.832	0./99																											
March	20.2	3			0.333	1.03	1.230		U o	2,630	2.399	2001	1.832	0.300	1.461	0.583	0.550																										
And	31.4				0.430	U:	U	U		1.403	1.323	0		0./96	1.589	0.490	0./9																										
April 1	213					U	U	U	U	0.513	0.2/8	U	U	0	0	0	0.001																										
May	31.4	2				U	U		0	0	Ū	0	0	0	0	0]			1																						
June	32.8	8			. 0	U	U	0	0	0	0	0	0	0	0	0																											
Allocation	70%	8			3.178	5.004	4.916	1.386	3.117	11.269	10.017	7.969	7.084	5.188	9.601	3.294	1.35	4	0 (3																							
Hydrology mi	odel input Jan	its - Note : giv Irrigation D	ven below is s liversions (ML	17 year average mont	hly rainfall d R C	lata and th Rainfall Cowra	erefore hys	drology cale	culations on	this page a	re based on I	hese avera	ge data		<u>.</u>									Aug Fi	arbes																Aug	Lake Car	gelligo
Varia L hudralau	00 CM	200000	200e 2	20101 20104	Jemaiong	Jan	Feb	Mar	Apr	May	Jun	JU	Aug	Sep	Oct	Nov	Dec	Jan	Feb	Mar	Apr	May	Jun	alloc	Jan	Feb N	Aar Apr	May	Jun	Jul	Aug	Sep ()ct	Nov D	ec Jar	n Feb	Mar	i Apr	May	Jun	alloc	Jan	Feb
Tear (hydroid	09.0%	29009	192/10	39191 9/546	64,205	58.1	44.6	49.5	46.2	49.4	50.8	52.3	50.9	51.7	57.1	51.8	57.2	58.9	44.7	49.0	43.7	48.7	50.3	69.6%	51.9	425 4	4.8 39.9	43.0	43.1	45.3	43.2	1.3 5	0.2	421 48	8 51.6	6 42.8	44.0	39.3	42.3	42.9	69.6%	39.7	34.2
Rainfall (MLA	ha)												-	-																													
	Effective	9	Rainfall 25%	Percentile	Effective Ex	cpected Ra	ainfall			Effective A	tual Rainfall																																
Month	rainfal	all Cowra	Forbes	Lake C. Hillstor	Zone 1	Zone 2	Zone 3	Zone 4	.lemalong	Zone 1	Zone 2	Zone 3	Zone 4	Iemalone				1																									
July	80%	\$ 26.25	23	145 13	0.210	0.184	0.116	0 104	0.116	0.418	0.362	0.299	0.061	0.280				-			1																						
August	80%	\$ 32.25	255	19.5 15.2	0.258	0.104	0.110	0.104	0.110	0.410	0.302	0.203	0.231	0.205			Industion I																										
Santembar	70%	K X	17.5	15 13.3	0.100	0.172	0.100	0.022	0.130	0.407	0.040	0.200	0.404	0.200			Ingauon	niciency																									
October	70%	κ 20 Κ 10	17.5	15 13.2	0.102	0.123	0.105	0.000	0.100	0.302	0.209	0.217	0.104	0.217				Ugr	t Heavy	1	-																						
November	70%	ະ 23 ເຊິ່ງງ	12 76	13 13.2	0.203	0.002	0.105	0.093	0.105	0.399	0.00	0.20/	0.252	0.26/			NU	/01	\$ 60%																								
December	70%	K 21	13.20	0 0	0.104	0.030	0.050	0.042	0.056	0.363	0.255	0.233	0.166	0.233			U	809	\$ 80%																								
Decemper	50%	N 21	13.63		0.105	0.076	0.055	0.040	0.050	U.200	0.229	0.186	0.152	0.186			Spray	903	% 75%																								
January	50%	8 D.D	۵	3	0.126	0.100	0.045	0.05	0.045	0.2%	0.258	0.199	0.155	0.199			Trickle	909	% 90%																								
recruary	50%	k 12	11	b .	0.060	0.056	0.030	0.015	0.030	0.224	0.214	0.172	0.131	0.172																													
March	70%	\$ 14	13	5.25	0.098	0.091	0.037	0.035	0.037	0.343	0.308	0.256	0.219	0.256																													
April	/0%	\$ 15	14	8.25 6	0.105	0.096	0.058	0.042	0.058	0.306	0.275	0.223	0.190	0.223																													
May	/0%	6 Z3	18	14.25 1	0.161	0.125	0.100	0.084	0.100	0.341	0.296	0.263	0.219	0.263																													
June	80%	6 25	24	16.25 13.2	0.200	0.192	0.130	0.106	0.130	0.402	0.343	0.299	0.263	0.299																													
Total					1.862	1.504	0.992	0.811	0.992	4.146	3.566	2.890	2.445	2,890																													
Crop Evapo-to	ranspiratio	ion requireme	ents (ML/ha)																																			1					
	Zone	1		A									Zone 2									Zone 3						Zone								Zone 5	- Jemalor	ng				-	
Marth	Asparagus	us Tomatoes	Sweet	Oats Wine-	Wheat	Canola	Maize	Lucerne	Lucerne	Lucerne	Perennial	Sub-	Apples	Wheat	Canola	Maize	Lucerne	Lucerne	Lucerne	Perennial	Sub-	Wheat	Canola	Maize L	ucerne Li	ucerne Per	ennial Sub	Wheat	Canola	Cotton	Maize So	rghum Lui	erne Lu	iceme Pere	nnial Sub	> Whea	at Canol	la Maizr	e Lucerr	ne Lucem	ie Lucernr	Perennial	Sub-
Modelh			com	grapes		0.0		seed	hay	pasture	pasture	clover					seed	hay	pasture	pasture	clover				hay p	asture par	sture clove	1				ł	ay pa	asture pas	ture clove	er			seed	hay	pasture	pasture	clover
July				U.5	0.25	0.2								0.25	0.2							0.27	0 22					0.2	0.22							0	27 0	122	-				
August				0.6/	0.51	U.46								0.51	0.46							0.58	0 53					0.5	0.53							0	58 0	153					
September				0.8	0.79	0.79		0.44	0.33	0.33	0.248			0.79	0.79		0.52	0.5	2 0.35	0.29		0.99	0.99		0.65	0.49	0 368	0.9	0.99				0.65	0.49	368	0	99 0	99	0.	66 0	65 0/	9 0.369	
October				0.4 0.4	0.91	1.04		0.93	0.9	0.9	0.675		0.54	0.91	1.04		1.08	5 1.0	5 1.03	3 0.77		1.2	1 38		1 39	1.37	1 028	1.	1.38	0.380		0.6	1.39	1.37	028		12 1	38	1	39 1	39 13	7 1009	
November		0.64	0.62	0.8 0.7	0.43	0.26	0.62	1.52	0.8	0.8	0.600		1.21	0.43	0.26	0.68	16	1.6	7 0.86	8 0.660		0.57	0.35	0.87	2.15	1.13	0 848	0.5	0.35	1.090	0.87	1.23	2 15	1.13	848	n	57 P	135 0	187 2	15 2	15 1	3 0.842	[]
December	0.89	9 1.58	1.4	1.0			1.4	1.45	1.22	1.22	0.915		212			1.69	1.76	17	6 1.49	1.118				1.86	1.93	1.64	1 230	1		2 130	186	1.14	1 93	164	230		- 0	- 1	86 1	93 11	93 15	4 122	- 1
January	1.09	9 1.9	1.82	1.0	1		1.82	1.49	1.24	1.24	0.930		2.39			2 23	1.8	1.8	1 15	3 1.14				2.49	203	17	1 275			2 550	2.49	172	203	17	275			2	19 2	m 1	ri 10	7 1 175	()
February	0.97	7 1.05	1.26	0.9	5		1.26	1.4	0.92	0.92	0.690	0.67	2.21			163	18	18	2 119	0.89	0.98			1.78	198	13	0.975 0	96		2 290	1.78	1 44	198	13	1975	196			78 1	09 1	09 4	3 0.075	0.00
March	0.74	4		0.74				1.21	0.98	0.98	0.736	0.98	1.59			1000	15	15	6 126	0 94	1.76				1.69	1 73	0.998 1	38		1 410		1.14	1.69	1 73	1009	1 38		1	10 1.	60 11	FQ 1.	0.0/5	1.30
April				0.4	1							0.27	0.89								0¥					1.00	0	35		0.440			1 00	1.30	1.550	136			1	uo I.	10 13	3 0.390	1.30
May				0.11 0.18	0.04	0.04								004	0.04							0.04	0.04				U	00	0.04	0.440						0	04 4	104					C.U
June				0.26	0.11	0.13								0.11	0.13							0.11	0.13					0.0	0.04							0	11 0	112					
Total	3.69	9 5.17	5.1	3.54 5.6	3.04	2.92	5.1	8.44	6.39	6.39	4.7925	192	10.95	3.04	2 92	6.79	10.15	10 1	8 77	5 877	2.49	3.75	364	7	11.00	0.00	677 7	60 2.7	264	10.2	7	6.62	11.02	0.00	6 72 -	200 2	20 1	1.1.1	7 11	02 (1)			

41

Appendix 3: Hydrology Data from DLWC's IQQM Model

August and January Simulated Allocation Percentages (* indicates where total offallocations from August to January added)

	C 71 B	ase Case	E 98	Rules	E 113	Rules	E 112	Rules	E 91	Rules	Elle	6 Rules	C 71A	Base Case	E 73/	Rules	E 131	Rules
	August	January	August	January	August	January	August	January	August	January	August	January	August	January *	August	January *	August	January
	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc
Year	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)
1901	120	119	100	100	100	100	100	100	100	100	100	100	120	127	100	104	100	100
1902	63	115	70	80	64	76	64	77	68	78	61	76	67	120	67	80	59	68
1903	5	4	0	0	0	0	0	0	0	0	0	0	5	7	0	2	0	0
1904	5	53	6	27	7	29	12	33	2	25	0	21	7	59	1	25	2	25
1905	24	29	18	18	18	18	21	21	17	17	18	18	26	31	18	19	17	17
1906	35	50	25	29	26	29	24	27	25	29	25	27	40	61	25	31	33	37
1907	13	100	6	61	5	61	6	62	4	61	4	61	13	106	5	64	6	63
1908	29	35	22	21	22	21	21	21	23	22	23	22	28	38	25	28	22	22
1909	4	35	7	22	10	25	10	25	8	22	10	25	4	38	10	26	6	21
1910	35	70	22	34	20	31	22	34	21	33	22	34	33	72	21	35	24	38
1911	41	66	32	36	31	35	30	- 34	29	33	28	33	42	73	28	35	30	34
1912	58	100	47	59	46	56	46	56	47	58	47	59	60	106	46	62	47	59
1913	47	109	34	57	33	55	33	55	33	57	33	57	49	115	35	59	31	56
1914	50	50	18	18	18	18	17	17	18	18	17	17	55	58	16	18	16	16
1915	20	19	23	22	22	21	22	21	23	22	23	22	20	22	24	26	23	23
1916	45	85	31	51	32	53	32	52	32	52	32	52	45	99	31	56	34	54
1917	90	119	55	100	54	100	55	100	56	100	- 55	100	92	136	54	105	57	100
1918	120	119	100	100	100	100	100	100	100	100	100	100	120	124	100	104	100	100
1919	109	119	100	100	100	100	98	100	100	100	98	100	111	126	100	100	100	100
1920	20	19	8	7	2	1	0	0	6	5	0	0	20	23	7	10	0	0
1921	20	50	2	33	2	31	1	30	2	33	1	30	3	60	2	37	2	33
1922	- 68	103	64	86	61	83	60	80	64	85	59	80	72	113	- 66	92	65	86
1923	66	98	48	55	49	57	49	55	49	55	46	53	70	118	49	60	44	53
1924	77	119	58	100	58	100	56	100	57	100	55	100	82	124	54	103	54	100
1925	65	119	78	97	73	93	69	93	75	88	69	93	70	126	78	98	68	79
1926	120	119	100	100	100	100	100	100	100	100	100	100	120	132	100	104	100	100
1927	120	119	100	100	100	100	100	100	100	100	100	100	120	125	100	103	100	100
1928	40	45	50	50	45	45	45	45	50	50	42	42	45	49	49	52	45	45
1929	120	119	100	100	100	100	100	100	100	100	100	100	120	125	100	102	100	100
1930	30	29	15	14	12	11	12	11	13	12	11	10	30	32	9	11	6	6
1931	1	53	0	25	<u>، -</u>	28	2	31	n	18	0	27	n	66	Û	31	ñ	21
1932	120	119	100	100	100	100	100	100	100	100	100	100	120	124	100	103	100	100
1933	114	119	100	100	100	100	100	100	100	100	100	100	115	127	100	103	100	100
1934	48	78	31	35	28	31	22	25	31	35	20	22	53	93	32	39	27	31
1935	50	119	39	100	39	100	38	100	39	100	38	100	55	127	37	104	38	100
1936	106	119	91	93	92	93	88	90	82	85	88	90	110	125	Q1	98	79	82
1937	117	119	73	82	74	80	73	79	68	74	72	77	118	125	76	88	64	74
1938	30	39	0	0	0	0	0	0	n	0	0	0	35	42	0	2	0	0
1030	1	35	0 0	20	n	21	0	23	n	21	0	21	1	30	1	24	0	18
1940	11	70	13	42	13	42	14	42	13	42	13	42	13	81	14	46	12	43
1941	5	4	4	3	5	4	5	12	5	12	7	г <u>г</u> 6	5	12	10	11	6	6
1042	0		8	7	g	7	0	0	J 0	0	40	11	J N	12	10	16	U Q	0
1043	5	63	18	42	10	A5	17	12	18	3	12	45	8	82	20	51	22	48
1044	120	110	100	100	100	100	100	100	100	100	100	100	120	128	100	102	2J 100	100
10/5	50	50	52	50	50	50	100	100	51	20	100	100	120	50	55	10J 67	50	52
1046	JU	20		0	0	30	4J 0	40	JI 0	30	4J 0	4J	-)-) -)-)	24	JJ	3	0	0
1047	م	20	0	0	0	0	0	0	0		0		2	24	0	J	0	0
1049	0	41	0	42	0	45	0	44	0	0	0	47	0	4	0	4	0	0
1740	77	412	00	42	02	40	04	44	02	92	02	97	0	40	07	4/	05	42
1797	11	113	30	100	30	100	39	100	30	100	33	37	10	113	3/	103	30	100

Appendix 3 -	Hydrology	Data	Continued
--------------	-----------	------	-----------

	C 71 Base Case		E 98 Rules		E 113 Rules		E 112 Rules		E91 Rules		E 116 Rules		C 71A Base Case		E 73A Rules		E 131 Rules	
	August	January	August	January	August	January	August	January	August	January	August	January	Åugust	January *	August	January *	August	Januar
	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc	alloc
Year	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)
1951	120	119	100	100	100	100	100	100	100	100	100	100	120	136	100	105	100	100
1952	120	119	100	100	100	100	100	100	100	100	100	100	120	138	100	105	100	100
1953	120	119	100	100	100	100	100	100	100	100	100	100	120	126	100	103	100	100
1954	104	119	100	100	100	100	100	100	100	100	100	100	104	124	100	101	100	100
1955	50	70	54	53	49	48	47	46	49	48	45	45	55	85	57	60	48	48
1956	81	119	67	100	66	100	67	100	56	100	65	100	82	132	66	104	56	100
1957	120	119	100	100	100	100	100	100	100	100	100	100	120	127	100	103	100	100
1958	70	70	85	85	78	77	73	72	81	80	73	72	75	79	82	85	76	76
1959	13	113	12	59	9	62	8	62	13	61	7	62	12	127	12	71	7	57
1960	120	119	100	100	100	100	100	100	100	100	100	100	120	140	100	105	100	100
1961	118	119	100	100	100	100	100	100	100	100	100	100	118	132	100	104	100	100
1962	116	119	100	100	100	100	100	100	100	100	100	100	116	131	100	104	100	100
1963	116	119	100	100	100	100	100	100	100	100	100	100	120	135	100	104	100	100
1964	120	119	100	100	100	100	100	100	100	100	100	100	120	131	100	104	100	100
1965	120	119	100	100	100	100	100	100	100	100	100	100	120	125	100	102	100	100
1966	45	54	54	53	47	46	48	47	50	50	45	45	45	60	52	55	47	47
1967	1	108	1	76	" 0	75	1	79	0	65	0	74	0	115	02	79	n	64
1968	49	70	52	59	50	57	52	61	43	52	47	55	54	75	52	62	44	53
1060	50	108	50	68	50	69	49	68	49	68	51	69	64	126	48	70	57	74
1070	120	110	100	100	100	100	100	100	100	100	98	100	120	133	93	104	100	100
1071	100	110	100	100	100	100	07	100	100	100	07	100	100	126	100	107	100	100
1072	100	110	100	100	100	100	100	100	100	100	100	100	100	120	100	102	100	100
1072	105	105	100	100	100	100	100	100	100	100	100	100	105	114	100	103	100	100
1975	100	110	50	100	100	100	44	100	100	100	100	100	54	107	50	104	100	100
1075	120	110	100	100	100	100	100	100	100	100	100	100	120	121	100	104	100	100
1076	120	110	100	100	100	100	100	100	100	100	100	100	120	100	100	104	100	100
1970	120	115	100	100	100	100	100	100	100	100	100	100	120	120	100	103	100	100
1070	120	115	100	100	100	100	100	100	100	100	100	100	140	121	100	104	100	100
1970	109	440	100	100	100	100	100	100	100	100	100	100	112	100	100	104	100	100
1979	70	70	00	70	74	74	70	70	00	70	70	70	75	04	70	00	72	72
1980	1 /0	//	00	/9	/4	/4	10	10	00	19	/0	10	10	01	19	00	13	/3
1701		20	U C	22	0	บ วง	U	U 74	U C	U 12	U	0 74	U 2	12	U T	1	C	ับ วง
1982	2	0	0	23	4	21	4	21	0 1	20	4	4	J 0	4J 1	1	20 C	0	21
1983		0	2	1	J 0	2	J 0	2	2	10	2	22	0	2	4	0	0	42
1984	400	20	0	400	U 400	្សា 400	U 400	ง1 400	U 400	10	0	JZ 400	U 400	JD 405	U 400	J4	400	10
1982	109	119	100	100	100	100	100	100	100	100	100	100	109	125	100	104	100	100
1986	50	119	00	93	00	91	00	04	00	93	5/	00	01	150	20	91	29	C0
1000	53	119	Jb	100	35	100	Jا دت	100	JD 07	100	32	100	55	120	JJ 00	104	29	100
1988	89	119	69	79	64	/4	b/	//	65	/6	63	/4	92	129	00	00 00	10	83
1989	8/	119	31	88	30	86	32	8/	32	89	31	86	88	125	32	89	JZ	90
1990	120	119	100	100	100	100	100	100	100	100	100	100	120	128	100	104	100	100
1991	120	119	100	100	100	100	100	100	100	100	100	100	120	131	100	104	100	100
1992	86	119	87	100	81	100	81	100	84	100	79	100	86	130	83	104	76	100
1993	113	119	93	100	84	100	83	100	88	100	82	100	113	129	85	104	79	100
1994	118	119	100	100	100	100	100	100	100	100	100	100	118	127	100	103	100	100
1995	80	79	91	91	84	84	81	80	89	88	81	80	80	89	89	93	85	85
1996	52	119	41	99	37	95	36	95	39	98	36	93	55	134	40	103	38-	99
1997	101	119	71	100	67	100	68	100	69	100	64	100	108	125	74	103	74	100
Mean	67.4%	88.3%	57.8%	69.6%	56.6%	68.9%	56.3%	68.8%	56.9%	68.7%	55.7%	68.3%	68.8%	97.0%	57.6%	73.1%	56.1%	68.3%
	4500	109/	20%	36%	30%	36%	30%	36%	30%	36%	30%	36%	45%	42%	30%	36%	20%	3696

Appendix 4: Results of LRMC's E98 and E116 (previous) flow rules analysis

	C 71 Base	e Case		E 98	8 Rules		E 116 Rules average January allocation 68.3%				
	average Ja allocation	anuary 88.3 %	aver	age January	y allocation 69.	6 %					
	Mean (\$)	SD (\$)	Mean (\$)	SD (\$)	Impact (\$)	Impact %	Mean (\$)	SD (\$)	Impact (\$)	Impact %	
Zone 1 Farm		6								1	
Gross Margin	86,890	13,355	83,946	15,257	-2,944	-3.4	83,589	15,480	-3,301	-3.8	
Net Farm Income	32,445	13,355	29,501	15,257	-2,944	-9.1	29,143	15,480	-3,301	-10.2	
Zone 2 Farm											
Gross Margin	158,070	15,092	152,786	14,731	-5,284	-3.3	152,302	14,772	-5,768	-3.6	
Net Farm Income	58,917	15,092	53,633	14,731	-5,284	-9.0	53,149	14,772	-5,768	-9.8	
Zone 3 Small Farm											
Gross Margin	109,854	8,127	106,895	7,155	-2,959	-2.7	106,580	7,232	-3,274	-3.0	
Net Farm Income	49,601	8,127	46,642	7,155	-2,959	-6.0	46,327	7,232	-3,274	-6.6	
Zone 3 Large Farm											
Gross Margin	228,794	10,242	225,915	10,580	-2,879	-1.3	225,901	10,772	-2,893	-1.3	
Net Farm Income	88,165	10,242	85,286	10,580	-2,879	-3.3	85,272	10,772	-2,893	-3.3	
Zone 4 Farm											
Gross Margin	429,397	27,751	421,298	30,824	-8,098	-1.9	421,036	31,407	-8,361	-1.9	
Net Farm Income	224,451	27,751	216,353	30,824	-8,098	-3.6	216,090	31,407	-8,361	-3.7	
Zone 5 Farm											
Gross Margin	147,009	25,319	140,568	29,766	-6,440	-4.4	139,734	29,865	-7,274	-4.9	
Net Farm Income	26,812	25,319	20,372	29,766	-6,440	-24.0	19,538	29,865	-7,274	-27.1	

